



Electromyography Monitoring Systems in Rehabilitation: A Review of Clinical Applications, Wearable Devices and Signal Acquisition Methodologies

Muhammad Al-Ayyad ^{1,*}, Hamza Abu Owida ¹, Roberto De Fazio ², Bassam Al-Naami ³, and Paolo Visconti ^{2,*}

- ¹ Department of Medical Engineering, Al-Ahliyya Amman University, Amman 19328, Jordan
- ² Department of Innovation Engineering, University of Salento, 73100 Lecce, Italy
- ³ Department of Biomedical Engineering, Faculty of Engineering, The Hashemite University, Zarqa 13133, Jordan
- * Correspondence: mayyad@ammanu.edu.jo (M.A.-A.); paolo.visconti@unisalento.it (P.V.); Tel.: +39-0832-297334 (P.V.)

Abstract: Recently, there has been an evolution toward a science-supported medicine, which uses replicable results from comprehensive studies to assist clinical decision-making. Reliable techniques are required to improve the consistency and replicability of studies assessing the effectiveness of clinical guidelines, mostly in muscular and therapeutic healthcare. In scientific research, surface electromyography (sEMG) is prevalent but underutilized as a valuable tool for physical medicine and rehabilitation. Other electrophysiological signals (e.g., from electrocardiogram (ECG), electroencephalogram (EEG), and needle EMG) are regularly monitored by medical specialists; nevertheless, the sEMG technique has not yet been effectively implemented in practical medical settings. However, sEMG has considerable clinical promise in evaluating muscle condition and operation; nevertheless, precise data extraction requires the definition of the procedures for tracking and interpreting sEMG and understanding the fundamental biophysics. This review is centered around the application of sEMG in rehabilitation and health monitoring systems, evaluating their technical specifications, including wearability. At first, this study examines methods and systems for tele-rehabilitation applications (i.e., neuromuscular, post-stroke, and sports) based on detecting EMG signals. Then, the fundamentals of EMG signal processing techniques and architectures commonly used to acquire and elaborate EMG signals are discussed. Afterward, a comprehensive and updated survey of wearable devices for sEMG detection, both reported in the scientific literature and on the market, is provided, mainly applied in rehabilitation training and physiological tracking. Discussions and comparisons about the examined solutions are presented to emphasize how rehabilitation professionals can reap the aid of neurobiological detection systems and identify perspectives in this field. These analyses contribute to identifying the key requirements of the next generation of wearable or portable sEMG devices employed in the healthcare field.

Keywords: electromyography; EMG instrumentation; tele-rehabilitation; signal processing

1. Introduction

Electromyography (EMG) is a neuro-muscular assessment method that involves detecting, monitoring, and evaluating biopotentials produced by motor units inside a muscular tissue during voluntary or involuntary actions. Two approaches may be identified based on the receiving sensor typology: intramuscular and superficial electromyography (sEMG) [1,2]. This enables the effective study of particular muscle activation, and has thus found several uses in medical investigations such as orthopedics, surgical procedures, nervous system studies, and gait and postural assessment [3–7]. EMG is also applied in



Citation: Al-Ayyad, M.; Owida, H.A.; De Fazio, R.; Al-Naami, B.; Visconti, P. Electromyography Monitoring Systems in Rehabilitation: A Review of Clinical Applications, Wearable Devices and Signal Acquisition Methodologies. *Electronics* **2023**, *12*, 1520. https://doi.org/10.3390/ electronics12071520

Academic Editor: Riccardo Pernice

Received: 21 December 2022 Revised: 28 February 2023 Accepted: 7 March 2023 Published: 23 March 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). risk prevention and ergonomic designs [8–11]. Sport-specific EMGs may provide a convenient point-of-care diagnostic test and a tool to improve sports performance [12]. Athletes use EMG to avoid muscular damage during performance evaluations [13]. sEMG can be utilized to develop muscle fatigue measurements in pre- and post-surgery monitoring and rehabilitation [14]. In recent years, EMG has grown in prominence for physical rehabilitation [15]; it also provides quantitative information on the muscle's myoelectric output and is widely used in neurorehabilitation research [16–18].

Surface electromyography allows recording of entire muscular biopotential signals from several muscle groups, evaluating the functional condition of a muscular area rather than simply a single motor unit [19]. The surface approach uses the characteristic of largescale electrical conductivity, which eliminates the influence of electrode proximity from the signal source on its form and character [20]. This approach enables the use of noninvasive electrodes, removing the pain and risk of monitoring. Surface electromyography detects and monitors the biopotentials generated when a neurological or electrochemical stimulus triggers muscle fibers. The responses include data on muscle activation, tone, and exhaustion, as well as recruitment and synchronization patterns [17,21–23]. EMG permits a more reliable interpretation of electrical events in the innervated muscles thanks to many years of study and continuous improvement of EMG signal recording technologies in detection and processing [24,25]. Simultaneously, a rapid technical growth of sEMG equipment has been observed, enabling more and more new chances to use this technology in various domains of medicine, predominantly in rehabilitation. For instance, identifying frames from sEMG data uses Gradient Boosted Regression Tree ensembles to predict wrist and finger kinematics, or novel algorithms to identify low-level hand movement by categorizing a single channel sEMG signal [26–30]. The rising use of sEMG in rehabilitation and physiotherapy, in the research field and clinical setting, suggests the need to present its applications, particularly in physiotherapy, where sEMG is commonly employed as a diagnostic and treatment tool [31–33].

Our review aims to provide an informative evaluation of the clinical applications of sEMG in rehabilitation and therapy by looking at the most current clinical applications of sEMG in rehabilitation and physiotherapy during the last decade. This paper comprises experimental studies and gives a summary of the treatment techniques employed and the findings and conclusions of the research listed. At first, an overview of methods and systems to acquire EMG signals suitable for tele-rehabilitation applications are discussed; then, the characteristics of EMG signals are introduced, along with the most common techniques and instruments to detect and process them. Afterward, an overview of EMG portable and wearable devices presented in the scientific literature is reported, and several commercial EMG wearable devices are reviewed. For both device categories, we focused on compact, unobtrusive, and wireless devices allowing simple and accurate detection of EMG signals and the derived parameters, such as amplitude and spectral parameters, time delay, correlation dimension, sample entropy, as well as RQA (Recurrence Quantification Analysis) parameters. The hardware and firmware of the described wearable devices were examined to reveal their strengths and limitations. Comparisons are provided in each section in order to outline the requirements of the next generation of EMG detection systems. The main strength of the presented paper is represented by the extensive reviews of the scientific literature, but above all, of commercial devices that are not analyzed in other similar scientific works. Comparative analyses are a strength of the presented work, enabling the reader to understand future perspectives of EMG sensors.

The main contributions of the presented review work are:

 A complete overview of methods and systems to acquire and analyze EMG signals for tele-rehabilitation applications. In detail, the discussion considers the main applications involving the EMG signals, such as neuromuscular rehabilitation, post-stroke rehabilitation, and sports rehabilitation.

- A detailed discussion about the signal processing techniques for EMG signals, as well as fundamentals about the structure and characteristics of the EMG signals, are provided, along with details on the architectures of EMG acquisition systems.
- A comprehensive and updated scientific literature survey about portable and wearable systems to monitor EMG signals applied to rehabilitation purposes. Comparative analyses are provided for determining the desirable features for the next generation of wearable EMG detectors.
- An up-to-date review of commercial wearable EMG detectors used for rehabilitation applications, and reporting comparative analysis to bring out their main requirements.

The remainder of the article is arranged as follows. Section 2 presents an overview of the main methodologies and systems to detect EMG signals applied to tele-rehabilitation applications. Section 3 discusses the main characteristics and processing techniques for EMG signals. Section 4 reports a survey of EMG portable and wearable devices. Finally, Section 5 reviews commercially available EMG sensors, as well as related comparative analyses are presented.

2. EMG Methods and Systems Applied to Tele-Rehabilitation Applications

The past decade has seen an emergence of advanced tele-medicine applications utilizing devices and computer technology [34]. As a relatively new and rapidly expanding field, tele-rehabilitation remains one of the most important applications. The benefit of tele-rehabilitation is that it reduces the cost both for healthcare organizations and patients compared to conventional inpatient rehabilitation or a face-to-face approach. Patients living in remote places can also benefit from tele-rehabilitation, which enables them to receive therapy remotely. Tele-rehabilitation has primarily been applied to physiotherapy [35]. It often takes the form of tele-monitoring, essentially efficient monitoring of physiological parameters in patients with chronic diseases, such as cardiovascular disease and oxygen levels [36]. Tele-rehabilitation enhanced treatment program versatility and eliminated the requirement for patients to go to sessions personally [37–39]. Additionally, patients with disabilities have reported physical and functional improvements [40–42].

Therapist-patient relationships developed through tele-rehabilitation were as good as those formed through in-person sessions [38,43]. Physiotherapy combined with telere-habilitation is an effective treatment option for musculoskeletal problems and physical illnesses. Furthermore, it helps healthcare professionals create a personalized physical training program for physical rehabilitation, improving patients' posture and mobility [44]. The examination and monitoring of neuromuscular problems are required in the rehabilitation sector to establish treatment aimed at developing and strengthening proper motor and sensory contractions. By acquiring these EMG signals, processing, and interpreting them, a powerful rehab tool for patients with severe impairments is obtained. Amplitude, timing, morphology, and spectral features of muscle activation can be expressed in various ways.

Numerous studies have been conducted internationally in the context of the COVID-19 epidemic to investigate the usage and feasibility of tele-rehabilitation, with differing data [41,45–47]. Persons with respiratory problems, for instance, embraced the online provisions and had better therapeutic results [45]. Similarly, despite some technical hurdles, persons saw tele-rehabilitation as practical and accepted it. Those persons noted disparities in service quality and favored traditional in-person therapy over tele-rehabilitation treatment [46]. Another study, on the contrary hand, discovered no statistically significant changes in patient satisfaction among patients who received in-person physical therapy compared to a patient who received remote physical therapy [47].

The possible clinical uses of sEMG sensors in rehabilitation medicine are described in the following sections, with an emphasis on (i) neuromuscular rehabilitation, (ii) stroke rehabilitation, and (iii) athletic rehabilitation.

2.1. EMG Applied to Neuromuscular Rehabilitation

An efficient rehabilitation process can restore some motor functions of damaged limbs in neurorehabilitation medicine and is an important aspect of clinical research [48]. Physicians need to assess their patients' physical and physiological health to determine if the exercise has an influence and, ultimately, to modify their follow-up rehabilitation program. The ability to identify impairments in the physician's clinical evaluation is aided by functional changes based on muscle activation data. With the ability to measure the muscle activation by EMG signals, electrodiagnostic medicine has become increasingly relevant and useful in neurorehabilitation, especially over the past four decades [49,50]. In [19], sEMG techniques are proposed to be utilized in neurorehabilitation. Firstly, the authors examine the use of EMG in neurological rehabilitation for assessing and treating muscle spasticity. This is due to EMG's capability to evaluate the changes caused by these abnormalities. The authors discuss a limited number of clinical applications. Manca et al. have conducted another study concerning surface electromyography utilization in neurorehabilitation [51]. In their study, they collected information regarding:

- (i) sEMG's present applications and therapeutic effects;
- (ii) professionals mainly concerned with sEMG;
- (iii) academic aspects;
- (iv) potential impediments and explanations for its seeming limited utilization in neurorehabilitation.

Different researchers in this field have proposed and discussed different aspects of sEMG use in neurorehabilitation, including the most recently developed applications, the educational, methodological, and technical characteristics, the possibilities of translation into clinical practice, and the possible benefits for patients and clinicians of this technique.

Neuro-rehabilitation employs surface EMG signals for (i) tracking neuro-muscular diseases, (ii) the avoidance of risks and disorders associated with the workplace, and (iii) observation and assessment of neuro-muscular state and healing progress in acutely ill patients. Valuable information regarding muscular activation patterns during motion and effort can assist clinicians in evaluating and providing a clinical assessment of both disability and functional changes [52–54]. Moreover, EMG and EEG (electro-encephalography) signals can be acquired through biosignal amplifiers to enhance the functionality of devices and systems used in brain-computer interface (BCI) applications. EEG and EMG can be used to predict the patient's imminent movements. It is not unusual to combine EEG and EMG in BCI applications to either predict as many movements as possible or to improve the prediction accuracy [55].

2.2. EMG Applied to Post-Stroke Rehabilitation

Among adults worldwide, stroke is a major cause of chronic disabilities [56]. Many stroke survivors suffer from hemiplegia, which hinders their ability to walk. Therefore, the rehabilitation of stroke patients is important to regain their motor coordination, muscle strength, and motor control [57]. From this point of view, exercises that enhance muscle activity and neuromuscular control are considered effective in motor rehabilitation [58]. EMG-based methods can help detect residual electro-muscular activity and, thus, assist in controlling exoskeletons, during the post-stroke period, for patients who can't generate enough torque for their joints [59]. Controlled neuromuscular electrical stimulation (NMES) in conjunction with electromyography has produced the best results for patients with stroke in clinical trials [60]. Accordingly, a systematic review was proposed by Monte-Silva et al. [61] about how EMG-NMES improves upper limb recovery following stroke. Another important study is provided by Hameed et al. [62]. The researchers show how assistive robotic devices can help patients with hand impairments perform everyday tasks and regain their ability to use their hands. In particular, they demonstrate that sEMG can control hand robotic devices, such as gloves and exoskeletons, for hand function recovery and enhancement.

Additionally, the researchers in [63] examined the possibility of EMG signals to detect the intention of hand or wrist extension movements, consequently triggering robot-assisted training. They present a comparison between detecting movement intention by an EMG sensor and a BCI using EEG sensorimotor rhythms. They concluded that therapy based on EMG devices has a legitimate and feasible approach to initiating robot-assisted training with a simpler interface and smaller dimensions than EEG-BCI systems.

2.3. EMG Applied to Sports Rehabilitation

Surface EMG is useful for assessing the status of skeletal muscles, which is important for muscular rehabilitation and physical exercises. Sport and rehabilitation scientists are increasingly using surface EMG as a research tool. The EMG is beneficial in the rehabilitation of athletes because it can be used to diagnose muscle impairment, detect incorrect muscle activation patterns, and evaluate treatment outcomes [12,64–66]. A better understanding of how to perform their tasks safely helps them avoid damage, since athletes need the proper usage of muscles and rapid detection of abnormal muscle patterns. sEMG's dynamic study of muscles is extremely useful in sports, notably for injury prevention [67–69]. For example, the measurement of the sEMG signal might develop the enactment of the exercise by assessing muscle activity and/or fatigue [70,71]. Fatigue analysis in the triceps brachii is an essential use of EMG in athletic rehabilitation. Based on sEMG, Hussain et al. [72] present an intriguing review of fatigue analysis in the human triceps brachii. Other contributions include post-operative rehabilitation when it is required for patients who have undergone rotator cuff surgery [73,74], and the study of particular muscle tissue interactions with external stimulation such as deoxygenation and exercise [12,75]. Trainers, coaches, and athletes can better understand the day-to-day need of athletes using wearable biosensors [76–78]. In [79], the author employed smart wearable sensing electronics and IoT technologies to create a sports rehabilitative tracking system. Sensors in this system collect and monitor biopotential signals, mobility orientation, skin temperature, and other vital signs. The experimental results reveal that the system can accurately monitor physiological parameter variations while providing immediate input and analysis. Additionally, the physiological data collected may be examined to assist clinicians in developing successful rehabilitation training regimens that provide extra features of the wearable system for EMG capture and processing to measure athlete performance. A proposed sEMG method has been tested by the authors in [80] regarding the capability to precisely and reliably determine muscle-firing waveforms during the isokinetic assessment of knee joint extensors and flexors.

3. EMG Signal Processing: General Considerations

The human physical activities produced by muscles, such as continuously pumping blood from the heart, are commanded and controlled by the brain. This gives rise to three primary bioelectrical signals: EMG, ECG (Electrocardiogram), and EEG. Detection, processing, and interpretation of these vital signs have been well documented in the literature [81–86] (Figure 1).

During limb motion, related skeletal muscles are neurologically stimulated and produce contraction, thereby generating action potentials that are picked up to form EMG signals [70,85]. As a result, EMG measurement has become an effective and widely used procedure in rehabilitation and management, as well as in monitoring physical activity and evaluating of muscle diseases. A well-designed electromyograph is used to capture high-quality EMG signals, and it is critical to ensure proper signal processing and EMG feature extraction [87].



Figure 1. Schematic of generalized biopotential measurement (main components) [87] (Reproduced with permission from [87], Elsevier, 2020).

3.1. EMG Signal Characteristics

EMG is an electrodiagnostic medical technique for evaluating and recording the muscular electrical signal produced by skeletal muscle movements [88]. The EMG signal represents neuromuscular activity by measuring electrical currents produced in muscles during contraction [89]. As a result, an EMG provides valuable information on muscle health and the function of motor neurons that carry electrical impulses to muscle fibers and allow them to contract. EMG signals have a frequency range of 20 to approximately 2000 Hz and an amplitude of roughly 50 μ V to 20 mV [90]. The biomechanics of human skeletal muscles or activity levels may be monitored and studied using the features analysis of EMG signals, which can provide body muscular activity information such as fitness, weariness, and stamina level, as well as a gesture. Muscle functioning during different levels of physical activities may provide important information about disability and functional changes [91]. EMG is employed in rehabilitative medicine, human-machine interface design, biomedical research, and various other applications, such as prostheses, as shown in Figure 2 [92,93].

EMG is used for physiological investigations, neurological disease monitoring, therapy planning, intervention evaluation, and control of prostheses and robotics [94]. Surface or internal electrodes are the two types of electrodes used in electromyography. Surface electrodes are used to track a muscle's overall activity, whereas nerve electrodes are used to disclose the electrical activity of a neuron. EMG signals show the status of limb muscle activity, representing skeletal muscle movement and nervous system control information, and are extremely valuable in stroke rehabilitation programs [91]. EMG signals may be used to identify and distinguish various limb movements, assisting in identifying and researching limb motions and their features [52]. Particularly, sEMG offers a non-invasive and thorough measurement of muscle activity that might be useful in movement analysis applications that need frequent evaluations or information on the activation patterns of various muscles [95]. Surface EMG, for example, might be a useful technique for quantifying progress and evaluating treatment results in sports, rehabilitation, and clinical

assessment [96]. Surface EMG is typically done on bigger, superficial muscles that are easily accessible. The surface EMG approach captures data on the amount of excitation from a broad region, which may comprise several distinct motor unit populations. Despite its widespread application and ease of implementation, surface EMG has disadvantages. For example, recording selectively from extremely deep muscles is not viable. In addition, cross-talk, an error source resulting from the vast pick-up region of the electrodes, may result in the recording of erroneous signals from deep muscles. Cross-talk must be handled carefully when analyzing an EMG signal, and can produce erroneous results. Techniques to limit cross-talk include strategically positioning electrodes on the surface, employing electrodes of the appropriate size, and maintaining a safe spacing between them. Because electrodes close to the innervation zone or tendon region may generate significant signal amplitude variation, it is essential to position them in the proper locations on the surface. During contractions, the movement of the muscle beneath the skin and the electrodes may also significantly impact the surface EMG signal [97,98]. Ag-AgCl electrodes are usually employed sEMG, including a conductive gel to reduce the impedance between the skin and electrode surface [87].



Figure 2. Schematic of the prosthetic hand control system presented in [93] (Reproduced with permission from [93], IEEE, 2012).

Internal EMG electrodes are rarely utilized due to their invasive nature. This technique is generally applied to evaluate deep muscles and those with a narrow cross-section. Unlike the implantation of surface electrodes, the insertion of electrodes requires particular knowledge and time. Thus, a professional operator (e.g., a neurologist, physiatrist, or physiotherapist) must place the electrode and monitor the muscle's activity. Intramuscular EMG is superior to surface EMG because it can detect a specific muscle's EMG signals under static and dynamic conditions with low interference. Two types of electrodes are common in internal EMG: a monopolar needle, constituted by a tip that acts as single electrodes, and the concentric needle, constituted by an inner core (active electrodes) inside an outer cannula as a reference electrode. In some circumstances, intramuscular EMG can be considered unneeded or excessively intrusive. The signal acquired from each electrode provides a very local representation of the muscle's activity. Since skeletal muscle inner structures vary, different locations must be considered to obtain a reliable analysis. Surface electromyographic signals are less spatially selective than intramuscular recordings due to the tissues' low-pass filtering impact on the sources (muscle fibers) and recording electrodes [99,100].

Similar to every biopotential signal, sEMG signals are nondeterministic, noisy, and complex; they also have small amplitudes and a frequency range. As a result, their acquisition is complicated. Noise from the electronic acquisition equipment, skin-electrode interface, and power lines all contribute to background noise. Therefore, a well-designed system is needed to enhance the acquisition and analysis of EMG signals. Such acquisition systems comprise electrodes, pre-processing stages (preamplifiers and filters), amplifiers, analog-to-digital conversions, power supply sections, and wireless transmission modules [101] (Figure 3).



Figure 3. Flowchart of electromyography acquisition system [101] (Reproduced with permission from [101], Springer Nature, 2019).

The amplification stage aims to increase the level of the differential signal between the electrodes while rejecting unwanted common noise. Then, the resulting signal is filtered and digitized with an analog-to-digital converter (ADC) for further analysis. In addition, the signal acquisition can occur in two different ways: the monopolar mode measures the differential voltage between an active electrode and an electrode-secluded marker located outside of a phasing region, whereas the bipolar mode measures the differential voltage between two electrodes. Furthermore, the system's wireless transmission module allows data to be collected, analyzed, displayed, and stored via an external PC. Furthermore, this module is essential to achieving a portable form of rehabilitation training.

The acquired EMG signal carries a wide range of desired and undesired information based on the surroundings, physiological, and instrumentation noise conditions. In particular, a pre-processing module composed of a low-pass and high-pass filter is necessary to remove the baseline and main frequency components from the EMG signal [102], and 50/60 Hz rejection, implemented with a notch filter, removes the harmonic noise generated by the power lines. The low-frequency cutoff of high-pass filters must be precisely defined because it is considered the main reason for the initial loss of amplitude in signals that change slowly, waveform misrepresentation, reducing the time to peak value, and originating artifacts [87]. A low-pass filter and high-frequency cutoff must also be o chosen to avoid reducing the amplitude and rise time of the informative component. After conditioning and processing steps, the EMG signal should be sent to a suitable data acquisition system to acquire, analyze, and/or store data. A digital value is assigned to the amplitude of the signal at predefined time points by a converter that discretizes the signal time and ampli-

tude. This procedure is required to perform further signal analysis for clinical diagnosis and research purposes.

In general, the EMG acquisition stage should meet the following substantial specifications [103]:

- Accuracy: many electronic elements, such as differential amplifiers, ADC converters, and others, are subject to intrinsic noise. The target is to minimize the noise in each element so that accuracy may be achieved.
- Sensitivity: pertains to the analog to digital resolution and, therefore, the total resolution of the device. This helps the medical staff control readings.
- CMRR: which stands for Common-Mode Rejection Ratio, and indicates the ability
 of a differential amplifier to reject signals common to both inputs. A high CMRR is
 essential in preventing 50–60 Hz power line interference.
- Input impedance: its compatibility is important in the selection of differential amplifiers and applications relative to the skin type and electrode interface.
- Input range: this specification applies to circuitry and the analog to digital converter, defining the range of EMG signal that can be amplified without saturating the amplifier. To acquire the complete signal, a greater input range is desired, but this necessitates an increase in signal resolution.
- SNR: signal-to-noise ratio measures the power of the desired signal relative to background noise.

4. EMG Portable Devices for Rehabilitation

Rehabilitation involves assessments and specialized training, but healthcare centers' limited resources often make this process challenging. For this reason, wearable technology represents a valid and important solution for objectively assessing and monitoring patients inside and outside of clinical environments. Using this technology, more detailed information about the impairment can be determined, allowing rehabilitation therapies to be identified [104]. The portability, low cost, and unobtrusiveness of wearable devices make this technology highly effective in tracking movements to improve neurologic or musculoskeletal care. As an added benefit, these sensors allow the evaluation of motor behavior, which is useful in compensatory motor recovery mechanisms, remote monitoring, and tele-rehabilitation [105–107].

Electrical biosignals can gauge the health and fitness conditions of the human body. In real-time e-health monitoring systems, biosignals such as ECG, EMG, and EEG are captured and analyzed to extract relevant and useful information for observation, diagnosis, and treatment [108].

As shown in Figure 4, these systems are generally configured as follows. The EEG, ECG, and EMG signals are extracted using electrodes placed on the subject and are acquired and processed by sensor devices connected to proper monitoring equipment. These systems generally have problems with slow data acquisition and transmission speeds, poor energy consumption, and cumbersome form factors that restrict their versatility.

Wearable devices monitor activity through two main processes: (i) acquiring, preprocessing, and managing data, and (ii) analyzing, classifying, and transmitting the data. Amplification and filtering are examples of signal pre-processing, while signal analysis comprises methods such as pooling or extracting crucial features used as a classifier test dataset (Figure 5) [54,109].

Smart sensors are often used in wearable devices to detect and monitor a collection of physiological characteristics to maintain a constant watch for the sake of diagnosis, treatment, and regulation [110]. The aging population's requirement for healthcare administration requires wearable medicinal products to collect individual health data promptly. During muscle exercises, electrical signal fluctuations create ECG and EMG, which are significant and widely used measures in healthcare management and rehabilitation regimens.







Figure 5. Transradial prosthesis based on the Myo wireless myoelectric armband proposed in [54]: (a) Adam's Hand prosthesis with its main embedded mechanical and electronical modules; (b) functional scheme of the employed electronic modules and related web application (Web-source: https://ietresearch.onlinelibrary.wiley.com/doi/10.1049/iet-smt.2018.5108, every aspect related to the human subjects' involvement has been taken into consideration by the authors in [54]) (Reproduced with permission from [54], John Wiley and Sons, 2019).

Zhao et al. presented a wearable device for upper limb rehabilitation [110], i.e., a robotic glove for assistance training that acquires EMG and ECG signals. In detail, these signals were acquired, pre-processed, digitalized, and transmitted through a Bluetooth Low-Energy (BLE) module to a remote receiver. Furthermore, a software platform for data processing was created by combining several instructions to show the captured electrophysiological data and reveal interest patterns. The EMG and ECG sensors, respectively, detect hand movements and changes related to these movements in a subject's physiological condition. The findings demonstrate that monitoring ECG and EMG signals can help the patient enhance upper limb improvement based on the treatment settings and needs of the users. In this work, wet electrodes were employed for detecting EMG, which can induce a change of contact impedance with the skin due to gel drying [111].

In [112], the authors present a wireless device to acquire and monitor physiological signals (i.e., ECG, EMG, PPG, and body acceleration). The system consists primarily of a portable device, a graphical user interface (GUI), and a software application for presenting the data on a computer or intelligent device. This system has eight measurement channels, a powerful microcontroller unit, a lithium battery, Bluetooth 3.0 data transfer, and a 2 GB

integrated flash memory. The results suggest that the developed device can help clinicians and scientists collect required physiological signals, exploiting its ability to acquire real-time data.

Similarly, in [113], the authors present a real-time and remote-control IoT system based on EMG and inertial signals. Using the Myo-band on an arm, a user may remotely manage domestic utilities (lights, room heaters, air-conditioners, ventilators, etc.) through eight gestures (Figure 6). The IoT system comprises four parts: sensing devices, gateway, cloud servers, and smart devices. The Myo armband collects EMG signals and motion-related inertial data, forwarding them via Bluetooth toward the gateway (Intel UPS-GWS01), which filters and extracts information from data and sends them to the cloud server. For managing electrical appliances and establishing a wireless connection to the home gateway, ESP8266 was employed. The tests demonstrated that the proposed system could achieve high accuracies, such as 100% for basic gestures and 90% for challenging ones.



Figure 6. Architecture of the EMG-based control system proposed in [113] (Reproduced with permission from [113], IEEE, 2021).

System latency linked to the internet network load could be a problem, heavily influencing the performance of the developed system.

Park et al. present an energy-efficient integrated circuit using a 128-channel Δ modulated $\Delta\Sigma$ analog front-end (Δ - $\Delta\Sigma$ AFE) for 1024-channel neural recording microsystems [114]. Platform components include eight multi-shank neural probes connected to individual AFEs (analog front-ends) based on a modular architecture using 128 channels (Figure 7). A spectrum equalization scheme was implemented to reduce the amount of area and energy consumed, taking advantage of the inherent spectral properties of neural signals (the bulk of the energy is found in the low frequencies). Δ - $\Delta\Sigma$ AFEs were designed to obtain the following features: the single-channel AFEs consume 3.05 W at 0.5 and 1.0 V from an area of 0.05 mm² with a 63.8-dB signal-to-noise-and-distortion ratio and a 3.02 noise efficiency factor.

Figure 7. Proposed IC in [114]: Conceptual diagram of a 1024-channel parallel-recording platform assembled with the proposed 128-channel AFEs, interposers, silicon caps, and 128-channel multi-shank probes, and a photograph of the fabricated module on the top of a U.S. penny (Reproduced with permission from [114], IEEE, 2018).

In [115], a multi-channel data acquisition system is described to record bio-electrical signals, including EMG. An eight-module front-end acquisition system is coupled with a synchronization module that ensures reliable synchronization between all acquired signals. A separate universal serial bus data link connects each front-end acquisition module to the computer. Using an external clock, it synchronizes with other modules, providing the microcontrollers with a time base. It is possible to analyze movements in real-time with a synchronization error lower than 10 μ s. Additionally, each AFE relies on the highly integrated ADS1299 chip, containing analog filters and an eight-channel ADC for digitalizing bipolar signals simultaneously. As a result, the proposed system can record up to 64 bipolar channels in real-time. In the end, raw data are analyzed and saved on a personal or single-board computer. However, the device presented in [115] does not have a wireless transceiver on board the acquisition board, and therefore requires a wired connection with the processing section (PC or Raspberry Pi board), limiting the use of the device in the acquisition of EMG signals in daily activities.

Tran et al. introduced their most recent contribution in [116]. The author developed a four-channel neural recording AFE integrated circuit (IC) featured by high power efficiency and low-noise. A low-noise amplifier (LNA), a programmable gain amplifier (PGA), and buffers made up each front-end channel. A 4-to-1 multiplexer (MUX) and an ADC accompany the four-channel AFE were used to acquire the four channels sequentially. The system had a programmable gain ranging from 45 to 63 dB and a 10 kHz operative band. The characterization demonstrated that the developed four-channel neural recording AFE produced 3.16 µV_{RMS} input-referred noise, a 2.04 noise efficiency factor, a 4.16 powerefficiency factor, and a 2.82 µW power consumption for each channel, considering a 1 V supply voltage. D.J. Piccinini et al. reported a versatile and wearable device for collecting and wirelessly transmitting biological signals [117]. This system relies on ane ADS1294 Medical AFE and CC3200 MCU (microcontroller unit, manufactured by Texas Instruments), customized for various signals such as ECG and EMG. The resulting solution is lightweight and powered by two Li-ion batteries. The test results showed that the developed device is very promising in terms of size, physical reduction, wireless transmission resilience, and data collecting and processing dependability. The device relies on an SoC (C3200, manufactured by Texas Instruments) with an embedded WiFi transceiver to transmit data to a host device where the data is processed. This limits the use of the device in areas where a WiFi hotspot is required.

Sarker et al. presented another application in [118], i.e., a compact wearable device for acquiring biosignals. Real-time data wireless transmission and minimal energy usage are its distinguishing features. The device is set up to record ECG and EMG signals over eight channels with a 24-bit resolution per channel and 500 SPS (samples per second) sampling rate. The system was presented as an example of prospective integration in an IoT-based system (Figure 8).

Figure 8. Device architecture proposed in [118] (Reproduced with permission from [118], IEEE, 2017).

The size of the presented device could be further reduced to enable integration into wearable devices, and the firmware further optimized to ensure better energy efficiency, extending its autonomy.

Mazzetta et al. presented a stand-alone wearable sEMG device for real-time monitoring of muscle activation [119]. This device can detect muscle activation potentials and, owing to an integrated low-power microprocessor, can perform entire real-time data processing. The system is designed for low power consumption, compactness, and energy independence, allowing it to collect patients' vital diagnostic data sets daily. Furthermore, the results of testing the system show that it had a specificity and sensitivity of over 87 percent and 82 percent in detecting correct activity time, respectively, with the added benefit of being wireless and comfortable to wear.

Another contribution in [120] describes 3D-printed smart glasses with bone vibration sensors and electrodes for electromyography (EMG) installed at the frame (Figure 9). To measure the EMG, a Bitalino EMG sensor was employed, using flexible fabric electrodes. A 64 μ m PET monofilament was woven in the warp and weft directions to create the electrodes. To record Temporalis muscle activations, two stripes were applied to the right temple's ear bend and temple end of the eyeglasses frame. EMG signals were acquired at 1 kHz and high-pass filtered at 10 Hz to eliminate baseline drift. The test results demonstrated that the developed smart glasses obtained a high SNR (15–20 dB) appropriate to identify chew activities. The main problem of the solution presented in [120] was maintaining contact between the electrodes and the user's skin, especially with the user in motion, a potential source of artifacts on the acquired signals.

For the capture of bioelectric signals in portable systems, compact and low-noise (AFEs) are becoming crucial. A low-power, multi-modal AFE for wearable health monitoring sensors is presented by Kim et al. [121]. It is based on CMOS (Complementary Metal Oxide Semiconductor) technology, revolutionary system architecture, and large-scale IC design approaches. Three sensors to measure bio-potential (i.e., ECG, EEG, and EMG), photoplethysmography (PPG), and bioelectrical impedance analyzer (BIA) are incorporated for reducing size and power consumption. The findings revealed that high-quality AFE enables users to easily self-monitor various clinically important physiological indicators. As

suggested by the authors, several functional blocks could be shared among the bio-signal and BIA AFEs, such as anti-aliasing filters and ADCs, further reducing the used area.

Figure 9. Prototype of the 3D-printed smart glasses, presented in [120] (**a**) with integrated bone vibration sensor and EMG electrodes applied on the right temple (**b**). Smart eyeglasses worn by a user (**c**) (Web-source: https://dl.acm.org/doi/10.1145/2971763.2971799, every aspect related to the human sub-jects' involvement has been taken into consideration by the authors in [120]) (Image courtsey by the authors, R. Zhang and O. Amft [120]).

In [122], the authors present a novel AFE featured by three properties: input impedance's dependency on voltage, bandpass amplification, and stray capacitance lowering utilizing capacitively coupled electrocardiogram (cECG) and capacitively coupled electromyogram (cEMG). The AFE characterization demonstrated that it could achieve a good balance of sensitivity and stability in capacitive biopotential measurements (CBMs). As a result, it is a more flexible solution than the traditional voltage followers employed in CBMs (Figure 10).

Figure 10. Proposed AFE (BNArf) [122] (Reproduced with permission from [122], MDPI, 2020).

Non-invasive EMG and accelerometer signals have been obtained using a wearable wireless device introduced by Biagetti et al. [123] to monitor human activity during sports and physical activities, as well as in healthcare applications. This system employs several tiny, lightweight wireless sensing nodes to collect, analyze, and transmit motion-related body signals (medical and accelerometer). These were sent to one or more base stations over an ad hoc 2.4 GHz radio connection. A user interface was also created for accessing, recording, and interpreting data from a remotely controlled personal computer linked to the base stations via USB (Universal Serial Bus). Data recorded from many participants were utilized to develop and test an automated classifier to determine the performed exercise for evaluating the system's capacity to identify the user's activity. On four activities conducted by three participants, the automated classifier obtained a maximum accuracy of 85.7 percent using data collected from acceleration and sEMG signals.

Following the earlier designed application, Biagetti et al. [124] reported a wireless sensing unit for the real-time capture of bioelectrical signals such as EMG and ECG. This instrument was designed to provide a continuous stream of data suited for individual activity recognition, motion tracking, and technology-assisted support for people with mobility or intellectual impairment (Figure 11). Up to three separate electrophysiologic channels, each of which has 24 bits of resolution and a sampling rate of up to 3.2 kHz, could be achieved using six electrodes. Furthermore, a BLE wireless connection was utilized to contact a wide range of consumer equipment. In particular, this work looked into the data rate restraints specified by these devices suggesting a technique achieving maximum available bandwidth and the transmission dependability.

Figure 11. Sensor node architecture with different power supply subsystems highlighted and employed for reducing the power requirements and the data interfaces. When the general-purpose input/output (GPIO) line is active, the ADS1293 collects the analog signals and transfers them to the SoC via a serial peripheral interface (SPI). Using inter-integrated circuit (I2C) communication, the LSM6DSO inertial measurement unit (IMU) provides data from it to the MCU [124] (**a**). The electrodes and electronic section are positioned on the top part of the right forearm, connected to the biceps brachii, triceps brachii, and deltoideus medium (highlighted in yellow, orange, and blue markers, respectively) (**b**). The image on the right below, which depicts the enclosure internal with the battery atop the circuit board, clearly illustrates the mechanical axis' alignment (**c**). When the device is put on, axes are positioned such that the Y is parallel to the arm and oriented downward at resting, the X is parallel to the sagittal plane and faces forward, and the Z axis is parallel to the coronal plane and faces the arm. (Web-source: https://www.mdpi.com/2079-9292/9/6/934, every aspect related to the human subjects' involvement has been taken into consideration by the authors in [124]) (Reproduced with permission from [124], MDPI, 2020).

The device's design and electrode configuration can be further optimized to make it more discreet and compact. In addition, some processing (filtering) sections may be integrated into the device's firmware, limiting post-processing.

Xian Li and Ye Sun offer a wirelessly non-skin contact button-like wearable system for long-term tracking of several bioelectric signals (ECG, EMG, and EEG) [125]. For noncontact sensing, this system relies on the analog front-ultra-high end's input impedance. The system is powered by a 150 mAh rechargeable Li-ion battery and comes in a 24.0 g 3D printed compact box with dimensions of 39 mm \times 32 mm \times 17 mm. A power control circuit is incorporated to offer a dual power supply to feed the operational amplifiers. Several movement patterns with various fabric types were used to assess the system's effectiveness, and the experimental results suggested that long-term bioelectric tracking is possible without disrupting everyday tasks.

D. Velumani et al. proposed an IoT-based tool for diagnosing bruxism, detected by measuring the activity of the masticatory muscles using sEMG [126]. This project's primary objective was to gather biosignals using an EMG sensor by placing surface electrodes over the cheeks and measuring muscle activity to identify signs of bruxism and its performance, as well as other metrics such as pulse rate. An ESP32 is the core of the developed wearable device to which EMG and HR sensors are interfaced. The patient is warned when the RMS (Root Mean Square) of the EMG signal overcomes the threshold value by sending an alert message to an IoT cloud platform, where the information is wirelessly saved and used for telediagnosis at any time. In our opinion, the main problem of the proposed solution is the stability of the contact between the EMG electrode and the patient's cheek, which could deteriorate due to the user's movements during sleep, inducing artifacts on the acquired signal.

The authors of [127] proposed a handheld device for EMG and other biopotential signal recording sections to assist in diagnosing and progressing many illnesses. The signal processing section removes baseline fluctuation (0.1–0.5 Hz) and (50/60 Hz) interfering components, and the processor generates control signals to set the AFE in two modalities: low noise–large CMRR and average noise–moderate CMRR modes. A signal filtering section was developed to pass the interest frequency band and reject the unwanted components. A Successive Approximation Register (SAR) DAC-digital to analog converter receives the control signals from the section above. The resulting biosignal elaborating section has a total size of 33.005 μ m² and power usage of 0.382 mW, thanks to Spartan-3E FPGA (Field Programmable Gate Array) and 0.18 μ m CMOS TSMC technology.

Lee et al. describe a new wireless ExG sensing tag with different channels for electrophysiological data acquisition (PSA) for collecting biopotential data [128]. In addition, a combined microprocessor system-on-chip (SoC) and a BLE transceiver was integrated for instant detection and wireless communication. The developed system includes an AFE, a PGA- programmable gain amplifier, a reconfigurable Σ - Δ ADC, and a 32-bit RISC processor. High-performance computing is used in FFT (Fast Fourier Analysis) and entropy coding processors with direct memory access (DMA) to decrease dynamic power. This device is designed to be energy efficient, and continuous recording of ExG signals in medical diagnostic instruments may be accomplished in as little as 12 h using a 200 mAH battery.

An ECG and EMG recorder featuring versatile architecture is presented in [129] that supports wired and wireless body sensor networks. Various options can be configured regarding hardware parameters, signal processing, and recording (Figure 12). The proposed architecture is arranged into three primary levels: data acquisition, processing, and transmission. A programmed analog front-end ADAS1000 with five single-ended gain lines with variable gain has been built. A 24-bit resolution ADC with a customizable data rate of up to 128 kHz has also been developed. A 24-bit resolution analog-to-digital converter with a customizable data rate of up to 128 kHz has also been developed.

Figure 12. Flow diagram of the data acquisition layer proposed in [129] (Reproduced with permission from [129], Elsevier, 2016).

The suggested system might be improved by automatically adjusting the remote recorder's software by the server by the reported diagnostic quality, allowing some firmware blocks to be tailored according to the target device.

In [130], researchers describe an interference-oriented biosensor front-end integrated into an ASIC (Application Specific Integrated Circuit) for wireless body sensor nodes and implanted biomedical instruments. The ASIC is made in a 0.18-micron CMOS processing and can be reconfigured to handle various bioelectrical signals, with high-pass and low-pass critical frequencies of 0.5–300 Hz and 150 Hz–10 kHz, accordingly. For the swapping 10-bit SAR ADC, an antialiasing filter is also provided. The AFE has a programmable gain ranging from 38 to 72 dB. A power control unit supplies the entire device with power, various voltage, and bias current levels. The AFE and ADC dissipate 5.74 μ W and 306 nW, respectively, and the device has 2.98 μ V_{RMS} input-referred noise, a 2.6 noise efficiency factor, and a 9.46 power efficiency factor. The AFE covers an area of 0.0228 mm².

A musculoskeletal rehabilitation evaluation (MSEva) system is proposed in [131] to assess neuromuscular activities using EMG signals, able to recognize five rehabilitative exercises from EMG data. MSEva collects signal characteristics with the Wavelet Transform (WT) and then trains its models with the Long Short-Term Memory (LSTM). The system employs an LSTM model to detect if the EMG response to rehabilitative procedures is normal. In terms of accuracy, MSEva reaches an average of 94.37%, which is significant for directing neuromuscular rehabilitation.

Table 1 summarizes the main features and strengths of scientific works on electromyography wearable monitoring systems published in peer-reviewed journals in the last six years.

		0	0,		1	
Reference	Year	Number of Channels	Signal Acquisition	Sampling Frequency	Wireless Connection	Technical Features and Strengths
S. Zhao et al. [110]	2020	2 channels	STM32L15 (MCU) + Precision instrumentation amplifiers + BMD 101 (16-bit ADC)	N.A. ^a	BLE	 A wearable smart monitoring system for upper limb rehabilitation based on acquiring and processing the ECG e EMG signals. Wireless data transmission to a remote processing device to reduce wearable device requirements. Integrated software platform based on multi-thread technology for detecting fatigue (ECG) and training progress (EMG) using an adaptative strategy.
S. H. Liu et al. [112]	2019	8 channels	MSP430 MCU (12-bit ADC)	100 Hz	Bluetooth 3.0 (BTM-204B)	 Portable and wireless acquisition system to record physiological signals (ECG, EEG, EOG, GSR, and PPG). Real-time monitoring of the multiple biosignals through proper smartphone and PC applications
M. Nguyen et al. [113]	2021	8 channels	Myo Armband	200 Hz	Bluetooth	 Real-time and remote household utilities. Edge computing exploiting a local smart gateway. Processing supported by inertial data
SY. Park et al. [114]	2018	128 channels	128-channel Custom AFE (10.9-bit Δ-ΔΣ ADC)	800 kHz	No	 Energy and area-efficient 128-channel AFE, including a modular Δ-modulated ∑Δ acquisition system. 1024-channel for high-density detection of brain activity. Implementation of spectrum equalization scheme to exploit the spectral features of neural signals. Low power consumption (3.05 µW/channel @ 1 V supply) and area (0.05 mm²). High SNRD (63.8 dB) and low NEF (3.02) for noise suppression.
L. Tran et al. [116]	2021	4 channels	4-channel Neural recording AFE IC (10-bit SAR ADC)	1–10 kHz	No	 Amplification that can be adjusted from 45 dB to 63 dB. Low input referred noise (3.16 V_{RMS}) inside the 10 kHz bandwidth. Low NEF (2.04) and high PEF (power efficiency factor) (4.16) Low power consumption (2.82 W/channel).

Table 1. Summarizing table of wearable monitoring systems reviewed in the manuscript (sorted by publication year).

Reference	Year	Number of Channels	Signal Acquisition	Sampling Frequency	Wireless Connection	Technical Features and Strengths	
D. J. Piccinini et al. [117]	2016	N.A. ^a	CC3200 MCU + ADS1294 AFE (24-bit ΔΣ)	32 kHz	WiFi (CC3200)	 Wireless wearable acquisition system for detecting biosignals (EMG, ECG, movement, body temperatur Discreet and reliable in data acquisition, processing, and transmission. 	
V.K. Sarker et al. [118]	2017	8 channels	ATmega328p + ADS1299 AFE (24-bit ADC)	250–1000 Hz	Bluetooth 2.0 (HC-05)	 Lightweight and wearable monitoring system gathering multiple bio-signals (ECG and EMG). High resolution (24 bit) and sample rate (500 samples per second). Extended energy autonomy (13.6 h with a 1700 mAh). 	
I. Mazzetta et al. [119]	2018	Differential 1 channel	Bio2Bit	≥4 kHz	Bluetooth 4.0	 Wearable sEMG system for real-time detection of muscle activity. Fully embedded system for the real-time elaboration of sEMG signals. Wearable, compact, and ubiquitous. Low power consumption (26 mW). High specificity and sensitivity in detecting the activation times (87% and 82%, respectively). 	
R. Zhang et al. [120]	2016	1 channel	Bitalino EMG	1 kHz	No	 Compact and low-power wearable and wireless system for continuously monitoring EMG signals and bone vibration. Fabric-based flexible electrodes applied on the temple ear bend and temple end. High SNR on EMG acquisition (15–20 dB). 	
I. Kim et al. [121]	2016	N.A. ^a	Cyclone IV FPGA + Custom AFE (10-bit SAR ADC)	1.10 MHz	No	 Low power and multi-modal AFE for monitoring PPG, BIA, and biopotential. Small covered area (2.5 mm × 2.5 mm) and power consumption (0.4 mW @ 1.2 V supply voltage). 	

Table 1. Cont.

Reference	Year	Number of Channels	Signal Acquisition	Sampling Frequency	Wireless Connection	Technical Features and Strengths	
H. Nakamura et al. [122]	2020	N.A. ^a	Custom AFE to capacitive biopotential measurements (CBMs) (16-bit ADC)	1 kHz	No	 Novel AFE architecture featured by: voltage-dependent input impedance, band amplification, and stray capacitance reduction. Suitable for capacitive biopotential measurements (CBM). Improved SNR 	
G. Biagetti et al. [123]	2018	1 channel	sEMG sensing nodes (12-bit ADC)	2 kHz	2.4 GHz radio link	 Wearable sensor node to detect and wirelessly transmit signals related to body motions (sEMG and inertial). Discreet and cost-effective Custom software platform for gathering and processing the acquired data to recognize human activity. 	
G. Biagetti et al. [124]	2020	3 channels	nRF52840 MCU + ADS1293 AFE (24-bit ADC)	3.2 kHz	BLE	 Wearable sensor for continuous monitoring of the bioelectrical (EMG and ECG) and inertial signals for activity detection. High resolution (24-bit) and sampling rate (3.2 kHz). Reliable wireless data transmission (BLE) toward a host device (smartphone or tablet), which acts as an aggregator. 	
X. Li et al. [125]	2017	N.A. ^a	ATmega328p (10-bit ADC)	N.A. ^a	2.4 GHz radio link (nRF24L01)	 Wearable wireless sensor for real-time and long-term monitoring of biosignals (EMG, EEG, and EEG), using multi-layer cloth electrodes without direct contact with the skin. Compact (39 mm × 32 mm × 17 mm) and lightweight (24 g). 	
D. Velumani et al. [126]	2022	1 channel	ESP32 (12-bit SAR ADC)	N.A. ^a	WiFi (ESP32)	 IoT wearable system for diagnosing bruxism. A cloud platform supports the device operation storing event when the RMS value of the EMG signal overcomes a threshold. 	

Table 1. Cont.

Reference	Year	Number of Channels	Signal Acquisition	Sampling Frequency	Wireless Connection	Technical Features and Strengths
B. Senapati et al. [127]	2017	N.A. ^a	Spartan-3E FPGA + Custom AFE (16-bit SAR ADC)	N.A. ^a	No	 Biopotential processor for acquiring ECG, EEG, EMG, and EOG signals. High resolution and capability of rejecting baseline wander and powerline interference (50/60 Hz). Two operative modalities: low noise-High CMRR and average noise and average CMRR modes. 33,005 µm² total area and 0.38 mW power consumption.
S.C. Lee et al. [128]	2016	8 channels	Open RISC 1200 MCU (12-bit ADC) + Custom AFE	N.A. ^a	Bluetooth 4.0	 Wireless ExG sensor tag for acquiring multiple physiological signals. Low battery consumption (12-h recording time)
Augustyniak et al. [129]	2016	Single-ended 5 channels	PXA-270 CPU + ADAS1000 (24-bit ADC)	500 Hz	WiFi	 Multi-purpose BSN architecture (wired and wireless) to detect physiological signals (e.g., ECG, EMG). Fully reconfigurable from the point of view of hardware configuration, data elaboration, and software management.
H. Bhamra et al. [130]	2017	N.A. ^a	ASIC 9.1-bit SAR ADC	4 kHz	No	 Integrated AFE for wearable and implantable devices, reconfigurable to acquire different biosignals. Amplification gain from 38 dB to 72 dB. Low noise (2.98 µV_{RMS} input-referred noise, 2.6 NEF, and 9.46 PEF). Low power consumption (5.74 µW for the AFE, 306 nW for the ADC).

Table 1. Cont.

^a N.A.: Not Available.

5. Commercial Wearable Devices

Wearable medical systems are designed to collect and process medical information regularly, rapidly diagnosing abnormal indicators and enhancing illness monitoring and management [132]. Wearable modified electromyography systems are commonly accessible in the market and are developed and manufactured to meet various requirements. The following sections present an overview of commercial biosignal sensing devices for vital signs in this context. These devices were chosen because they are the most used devices in diagnosis and have comparable qualities that may be measured [133].

Biometrics Ltd. (Newport, UK) provides a variety of monitoring devices to acquire analog or digital data from various sensor typologies; different device categories are available, namely mobile, handheld, and lab versions. The proposed wearable devices enable a complete range of motion without needing cables [134]. Biometrics Ltd. provides non-invasive wireless electromyography sensors and surface EMG amplifiers (Figure 13). They are available in 2-, 4-, 8-, and 16-channeled variants for acquiring EMG signals with non-invasive, compact, and light detectors, enabling secure and effective myoelectric measurements at a distance of up to 30 m from the medical staff who analyze the data. The major characteristics of these detectors are operating bandwidth ranging from 10 Hz to 250 Hz up to 5 kHz, and sensitivity ranging from +/60 mV to +/6000 mV for fullscale peak-to-peak readings [135]. Wearable devices are constituted by both sensors and instrumentation systems for stationary and non-stationary measurements in a medical environment, an academic institute, or any distant point such as an office, work, or house.

Figure 13. Biometrics Ltd. EMG sensors and systems are used extensively in the fields of ergonomics, sports science, and medical research, where measurements are commonly taken for a wide range of applications. The surface EMG sensor LE230 can be positioned in different body positions, such as arms (**a**) and legs (**b**) (Web-source: https://www.biometricsltd.com/surface-emg-sensor.htm arms (**a**), https://www.medicalexpo.com/prod/biometrics/product-123702-880665.html legs (**b**), every aspect related to the human subjects' involvement has been taken into consideration by Biometrics Ltd.) (Image courtsey by Biometrics Ltd., Newport, UK).

Several sensors are available from Shimmer Co. (Dublin, Ireland) for measuring various parameters [136–138], including configurable digital front-ends for EMG data acquisition and evaluation [137] (Figure 14a). They manufacture surface transducers to monitor muscle activity. By a shared referenced electrode, the wireless EMG system offers multiple information channels. Besides EMG, ECG signals can be acquired, but the two measurements cannot be made simultaneously. A 450 mAh rechargeable Li-ion battery powers the Shimmer3 EMG unit, which includes an MSP430 microprocessor, a Bluetooth

transceiver (RN-42), and an in-built 8 GB micro-SD card. Since the Shimmer 3 EMG sensor has a centralized acquisition and processing unit to which the EMG probes must be connected (Figure 14a), it can be uncomfortable and hinder the movements due to the presence of cables.

Figure 14. Applications of the Shimmer3 EMG (**a**) and FreeEMG (**b**) sensors on the legs and shoulder for monitoring body movements (Web-source: https://www.biometricsltd.com/surface-emg-sensor. htm (**a**), https://www.medicalexpo.com/prod/biometrics/product-123702-880665.html (**b**), every aspect related to the human subjects' involvement has been taken into consideration by Shimmer Co. (Dublin, Ireland) and BTS S.p.A. (Garbagnate Milanese, MI Italy), respectively) (Image courtsey by Shimmer Co. (Dublin, Ireland) and BTS S.p.A. (Garbagnate Milanese, MI Italy), respectively).

Both previous wearable devices (Biometrics and Shimmer 3) were employed in [139] for validating proposed EMG sensors. The experimental results demonstrated that both commercial devices obtained a good SNR (\approx 20 dB). In contrast, the EMG sensor solution reached an SNR in the range of 11 and 18 dB according to the method used for SNR calculation. Biometric EMG sensors were involved in a scientific study to analyze the influence of related muscular co-activities (i.e., vastus lateralis and gastrocnemius) with knee flexor-extensor actions associated with the onset of Osteoarthritis (OA) [140]. The outcome focused on the postural efficacy of muscular activities in the stability problems of knee joint motions and demonstrates that the voltage amplitudes of EMG signals changed significantly with advancing age.

BioSemi Instrumentation Co. (Amsterdam, The Netherlands) offers the ActiveTwo bioelectric detection device for scientific purposes [141]. This system integrates up to a 256-channel DC amplifier, 24-bit ADC per channel, and active electrodes, which are thinner and lighter than previous models, with significantly improved low-frequency noise and input impedance specifications. Notably, the Active Two system offers up to 256 + 8 electrodes + 7 sensor channels in a single ultra-compact box, a battery-powered front-end with fiber optic data transfer, reliable readings without skin treatment, and improved digital resolution with a 31 nV LSB (Least Significant Bit) value, and a user-configurable sampling rate of 2, 4, 8, 16 kHz/channel. The system provides the user with a graphical LabVIEW application for controlling the different sensor parameters, making it suitable for detecting different biosignals, such as EEG, ECG, and EMG. The ActiveTwo EMG sensor is impractical for EMG monitoring during daily life, as the acquisition and processing unit is bulky and requires a large flat cable to connect the electrodes. In addition, the device has no transceiver for wirelessly transmitting the acquired data.

In [142], an ActiveTwo EMG sensor acquired and analyzed tiny intestine bioelectrical signals through flexible PCB electrode arrays, enabling up to 256 simultaneous recordings. The Falling-Edge Variable-Threshold technique was used to automate data processing,

resulting in 92% sensitivity and a 94% positive predictive value. The automatic production of animations and isochronal maps allowed for the visualization of slow wave propagation patterns.

The FreeEMG (manufactured by BTS S.p.A., Garbagnate Milanese, MI Italy) is a portable electromyography system that analyzes musculoskeletal activity in real-time [143] (Figure 14b). It is a surface EMG monitoring device with 4G technology that ensures data reliability, lightness, and comfort, thanks to the absence of wires and the smaller probes. Orthopedic and neurologic problems, pharmacological therapies, the progression of motor impairments, retraining and follow-up, and sports performance adaptation are all common uses for FreeEMG. This sensor was employed in [144] to explore the dynamic asymmetries of healthy individuals' lower limbs during gait adaptation on a split-belt treadmill. Ankle, knee, and hip joints, joint rotations, muscle lengthening, and SEMG were concurrently monitored, along with the produced sagittal power and work. According to the SENIAM (surface EMG for non-invasive assessment of muscles) recommendations, four sEMG probes were placed bilaterally on the bellies of the Tibialis Anterior, Gastrocnemius Lateralis (GaLat), Vastus Medialis, and Semi-Membranosus. The obtained results confirm that, on solid ground, the problematic leg "escapes" from load while walking with a limp, whereas in split gait, the quicker leg "escapes" from being pulled behind, the slower limb.

PLUX Wireless Biosignals Co. (Lisbon, Portugal) has created biosignal collection and monitoring solutions that incorporate wearable body sensors such as EMG and ECG, as well as wireless connection and software applications [145]. BITalino and Biosignalplux are two of these systems. The BITalino (r)evolution kit is a versatile board that includes all parts pre-connected and available to use outside the arrangement, as well as Bluetooth connectivity [146]. Its non-invasive EMG electrodes are created specifically for it. The bipolar design is perfect for limiting the noise on collected data, whereas the unfiltered output makes it suitable for both human-computer interface and biological applications.

Biosignalsplux is a wireless toolbox for collecting and analyzing high-resolution biosignal data [147]. It comes with a variety of wired and wearable sensors. For gathering muscular information, the Biosignalsplux EMG sensor is a high-performance bipolar sensor with minimal noise. The bipolar architecture of this sensor is suitable for acquiring low-noise data meant to detect muscle activity. The raw data generated is medical-grade, allowing it to be used for sophisticated medical biomechanics and athletics studies. The main features of this system are the usage of bipolar differential detection, availability of pre-processed analog output, high SNR, and medical-grade unfiltered output data; it is also tiny and simple to use. The muscleBAND is a portable single-channel EMG instrument for continuous muscle monitoring. It is a single-channel EMG sensor with a triaxial accelerometer and magnetometer for gathering muscle activity and motion data and a dual Bluetooth module. The integrated battery adequately feeds the device, ensuring long-term data streaming and allowing data collection with up to 16-bit resolution at up to 1000 Hz sampling rate.

The main limitation of Trigno Avanti sensors is the need to use a proprietary wireless station (Trigno Wireless System) for receiving data from the sensors using a proprietary RF protocol, which could be a problem when collecting data from different typologies of sensors. The BiosignalsPlux[®] device was utilized to track the shoulder muscle contraction pattern during five ADLs (activities of daily living) with various motor patterns [148]. The major goal of this article is to demonstrate the use of BiosignalsPlux[®] sensors to characterize the shoulder muscles' pattern of contraction while performing ADLs in healthy people. The results indicate that the pattern of shoulder muscle contraction varies depending on whether an ADL is aimed at the midline or the opposite side, and that various ADLs directed at the midline exhibit diverse behaviors.

Delsys Inc. (Natick, MA, USA) offers wearable EMG-based body motion tracking systems for scientific, medical, and teaching applications [149]. These technologies include scientific, handheld, light solutions, EMG electrodes, mobile platforms, and device integration software. Trigno Avanti Sensor is the most commonly used EMG detector

since it effectively detects muscle movement signals. It has patented technology, increased radiation efficiency, wireless design, adjustable EMG frequency choices, and integrated signal analyzing, and it is intended to operate with all Trigno units. It also enables bipolar EMG measurement within a very tiny size and weight. Trigno Research+ is an effective-performance system that makes EMG signal sensing reliable and simple while providing a comprehensive range of physiologic and musculoskeletal detection capabilities to facilitate complex studies and produce the best qualitative data. Trigno's proprietary RF protocol ensures sensor coordination and data transfer from Trigno sensing devices to a Trigno base station.

In [150], Z. Hu et al. analyzed badminton single-leg landing following an overhead shot, and demonstrated that a deficiency of biomechanical indicators of the knee indicates the risk of anterior cruciate ligament (ACL). Kinematic data were gathered through 13 infrared cameras with a 120 Hz sampling rate. EMG data were collected using the Trigno Avanti sensor, placed on the quadriceps femoris (rectus femoris), medial hamstrings (semitendinosus), lateral hamstrings (biceps femoris), medial gastrocnemius, and lateral gastrocnemius; the sampling frequency was 1200 Hz. The results suggest that during the single-leg landing in badminton, lower extremity muscle activity and knee kinematics and kinetics are correlated; thus, lower extremity muscle activity should be taken into account to create rehabilitation or injury prevention programs.

Table 2 compares the commercial EMG sensing devices previously discussed from the point of view of the additional detected signals, the number of EMG channels, dimension, weight, sampling frequency, electrode typology, availability of wireless connectivity, and suggested applications. In this way, different devices can be compared to highlight their potentialities and limitations.

From the presented analysis, it can be deduced that the ActiveTwo EMG monitoring device is the most complete and flexible solution for acquiring biopotential signals, including EMG signals, given its wide range of acquisition channels (280 channels) and high resolution (24-bit). Nevertheless, because this system is wired, it is not suitable for monitoring EMG signals when the user is moving. Likewise, the FreeEMG system represents an optimal solution in terms of compactness, discretion, and ease of use, given its small dimensions (27 mm \times 37 mm \times 15 mm), weight (14 g), and availability of wireless connection (WiFi), enabling its employment in every user condition. Furthermore, it guarantees good resolution (16-bit) and sampling rate (2 kHz) in detecting the EMG signals.

Device	Other Detected Signals	N° of EMG Channels	Dimension/ Weight	Sampling Frequency [Hz]	Electrode Typology	Wireless Connectivity	Suggested Applications
Biometrics wireless sensors [134]	Joint angle (electro- goniometers- optionally)	2, 4, 8, 16	N.A. ^a	500, 1000, 2000	Disposable sEMG electrodes with 4 mm snap	Yes (WiFi)	 Symmetry studies during gait Timing data in biomechanics Sport performance monitoring Neuro Rehabilitation
Shimmer3 EMG units [137]	ECG	2	65 mm × 32 mm × 12 mm/31 g	125-8000	Patented disposable EMG electrodes	Yes (Bluetooth)	• Gait, muscle, and posture disturbances analysis
ActiveTwo EMG unit [141]	EEG, ECG	280	N.A. ^a	200, 400, 8000, 16,000	Special silver-plated electrode-tip	No	• General bio-potential measurements for research applications
FreeEMG sensors [143]	Joint angle (electrogoniometers- optionally)	1	27 mm × 37 mm × 15 mm/14 g	≤4000	Standard with a clip connection	Yes (WiFi)	 Functional evaluation of gait analysis Sport biomechanics. Injury prevention and return to play. Cognitive and mobility rehabilitation.
Trigo Wireless EMG sensors [149]	Inertial data	4, 8, 16	N.A. ^a	2000	Silver electrodes	Yes (WiFi and BLE)	 Sport science Rehabilitation sciences Robotics Kinesiology. Speech pathology. Gaming-rehabilitation Motor control

Table 2. Comparison between the commercial devices previously discussed from the perspectives of additional acquired signal, number of channels, dimension, weight, sampling frequency, electrode typology, availability of wireless connectivity, and suggested application.

^a Not Available.

6. Discussion

Using and interpreting EMG signals in rehabilitative clinical follow-up is a valuable tool in identifying and monitoring muscles' healthy and unhealthy electrophysiological conditions. A significant amount of research has been conducted on surface electromyography. EMG signals provide measurable information about a muscle's biopotential signal. However, research outcomes and actual medical applications are conflicting regarding using sEMG. Several factors are responsible for this, including shortage of time and knowledge due, for example, to sensor locations and device configuration, medically unrelated sEMG device characteristics, primarily due to sEMG's restricted spatial resolution, and insufficient education and assertiveness in sEMG implementing technology. The obvious issues are technological in nature and include data analyzing and processing algorithms that do not immediately provide clinically useful data. Another difficulty is that certain equipment is difficult to operate.

Furthermore, the equipment's cost and the timed process to perform a study and acquire medically crucial results have detrimental effects. Concerning wearable applications, the main required features for a wearable sensor are minimal power consumption, portability, and reliability. Furthermore, portable equipment must be small and comfortable enough to be worn. Additionally, they must be equipped with a storage device and radio transmission to continuously monitor human activity in every condition. Numerous commercialized sensors are now flexible enough to meet these criteria, providing tailored solutions based on unique demands. Because of their tiny size and wearability, these sensors are more useful in patient control in hospital settings and tele-medicine scenarios.

The main issue that wearable EMG sensors may confront is that each time a muscle is utilized, the electrodes shift slightly about the underlying musculature. Electrode movement occurs during user activity due to stresses and limb positioning. A change in the limb's EMG characteristic (recording) as a result of such an electrode relocation can make analyzing the motions more difficult.

It is common practice to record an individual's EMG signal from their limb position when they are in a stationary position (sitting); however, in a real-world scenario, users must employ the device in various positions (walking, climbing stairs). On the other hand, the accuracy of EMG categorization is affected by subtle changes in limb posture. The same limb assists different muscular contraction forces throughout daily tasks. Therefore, the discrepancy in myoelectric signal pattern classification happens because muscle contraction force varies even when the same limb is targeted. Furthermore, the change in gravitational force caused by different leg postures causes the displacement of target muscles. EMG signal pattern variation is a result of these factors. Modern commercial sensors are limited mostly by their high cost and the system's complexity employed by a single subject, and they are not always simple to utilize. Frequently, these commercial sensors have complicated designs primarily centered on advanced technology with minimal emphasis on pleasant sensor-subject contact. An additional issue is that such devices are built using laboratory measurements, prototypes, and patient simulators, neglecting the factors and performing well in simulation but exhibiting flaws in real-world situations. Because the most recent sensor technology must fulfill the demands of physicians and patients, experimental models are now more vital in overcoming these restrictions. State-of-the-art EMG sensors should detect prospective diagnostic and therapeutic needs by highlighting relevant bioelectrical signals associated with certain muscular activities and pursuing the creation of solutions based on innovative detection systems. These new solutions must be able to track muscular activities from convenient places by employing innovative communication techniques. The equipment must also be constructed with these limits and needs in mind. In the near future, the sEMG technique will aid in collecting myoelectrical signals at home, proactive clinical procedures, and improving remote treatment and rehabilitation programs. As a result, the next stages will focus on creating smaller and cost-effective devices that a broad population will utilize.

Even once these studies are done, it may be a while before the discussed wearable systems or treatments are ready for commercialization. Developers of devices are concerned because they must spend a lot of money to bring their products to the market. It may not pay off to be the first to market with cutting-edge technology if the new product's window of opportunity for big profits is small. Bringing new technology into hospitals is a complex process that requires buy-in from both doctors and upper administration.

Despite efforts by some of the world's leading academic institutions to hasten innovation and the introduction of new goods to the market, progress is slowed by a lack of cooperation and exchange of ideas between the medical and engineering communities. A sustainable medical device business has emerged in developing nations, partly due to the reverse engineering of current items, to compensate for the scarcity of affordable devices.

The medical device R&D process and the subsequent commercialization of these products are extremely high-risk endeavors. It takes a long time and a lot of money to take an idea from the drawing board to actual use in the clinic. Early research is typically conducted in universities, whereas the rest of the process, including testing and manufacturing, is handled by private industry. Procedures typically take a long time and a lot of money to complete. Despite extensive ex vivo and in vivo testing, there is always a potential that a new product will fail, leading to major medical problems for the individual and financial devastation for the producer. Even if they face different health problems, the wealthiest countries still struggle with healthcare distribution and access. In theory, everyone in need should have access to transplantation and other life-support equipment such as dialysis and circulatory assist devices, but in practice, access is severely limited around the world. This exemplifies moral quandaries that are relevant to other costly implantable technologies as well.

The merger of medical and technical expertise may result in more rapid and targeted development and, as a result, more investment resources may be available. Early research and development can take place through consortia that include academics, industry, and government agencies, which can reduce investor reluctance to take risks. It is possible that streamlining clinical testing and taking a more consistent approach to the health technology assessment process will hasten the process of introducing cost-effective devices into the market and spreading their use. Alterations to existing patent rules and the application of those laws could lead to lower prices and greater levels of competition. Although decreased pricing would threaten the current business model that device firms use, increasing sales might be able to repay them financially by forming an open worldwide market. There is a high likelihood that the pricing of medical gadgets, which can be extravagant at times, cannot be maintained in the world's wealthy or low-to-middle-income regions. Many potentially life-saving medical technologies have the potential to become massproduced items at prices that are affordable. When dealing with medical technology, such advancement would make it simpler for the medical industry to stay true to the fundamental ethical values that guide it.

7. Conclusions

Electromyography is a clinical test to evaluate the condition of the muscular and nerve cells that drive them (motor neurons). EMG results can identify difficulties with nerve-tomuscle signal transmission, muscle dysfunction, or both. This article deals with the most generally utilized electrophysiological monitoring devices in the field of rehabilitation. It focuses on sEMG monitoring as one of the key platforms for assisting doctors, injured patients, and general people in rehabilitation. At first, an overview of the main EMG methodologies and systems applied to tele-medicine applications is introduced. Then, the characteristics of the EMG signals are discussed, along with the common approaches for conditioning and processing them. Afterward, a survey of EMG portable and wearable devices is reported in detail, many references from the literature being included in this study to demonstrate the relevance of EMG signal capture, processing, and tracking to muscular activity management. Many commercially applied EMG detectors are reviewed and compared to determine the most common characteristics of such sensors in EMG recording. The main requirements are low dimension and size, reduced power consumption, and the availability of wirelessly data transfer to ensure actual usage in rehabilitation.

To make the EMG-based motor intention prediction more practically applicable, significant attention should be paid to many joints while they are engaged in complex motion circumstances. In addition, as opposed to rehabilitation, more general applications may require the operators' left and right arms to move, and these movements may frequently have different trajectories.

Hence, figuring out how to use EMG signals to anticipate the movement intention of one arm while the other arm is simultaneously in motion, and how to predict the movement intents of both arms simultaneously, are useful in practice, hinting at a new research avenue.

Several approaches have been developed to integrate the EMG and the intelligence of robots to increase the overall performance of collaboration systems. These methods are seen from the point of view of shared control. To make up for the shortcomings of the EMG signals, systems that use shared control might be utilized. The concept of shared control was developed as a solution to address several issues, including the potential for hazardous circumstances and accidents, the imprecision of human control, and the exhaustion that can result from maintaining continuous control over a device. As a result of the limitations inherent to human control, an intelligent controller and a human controller may have some level of influence over the equipment being controlled. As a result, advanced approaches and procedures for shared control can also considerably improve EMG-based human-robot collaboration systems.

Author Contributions: Conceptualization, M.A.-A., R.D.F. and B.A.-N.; methodology, P.V. and R.D.F.; validation, P.V. and B.A.-N.; formal analysis, R.D.F., M.A.-A. and B.A.-N.; investigation, M.A.-A., P.V. and B.A.-N.; resources, P.V.; data curation, P.V. and R.D.F.; writing—original draft preparation, M.A.-A., H.A.O. and R.D.F.; writing—review and editing, R.D.F., P.V. and B.A.-N.; visualization, P.V.; supervision, P.V. and R.D.F.; funding acquisition, P.V. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Data of our study are available upon request.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Mills, K.R. The basics of electromyography. J. Neurol. Neurosurg. Psychiatry 2005, 76, ii32–ii35. [CrossRef] [PubMed]
- 2. Mizrahi, J. *Advances in Applied Electromyography*; IntechOpen: London, UK, 2011; ISBN 978-953-307-382-8. Available online: https://www.intechopen.com/books/359 (accessed on 20 December 2022). [CrossRef]
- Flaxman, T.E.; Alkjaer, T.; Smale, K.B.; Simonsen, E.B.; Krogsgaard, M.R.; Benoit, D.L. Differences in EMG–moment relationships between ACL-injured and uninjured adults during a weight-bearing multidirectional force control task. *J. Orthop. Res.* 2019, 37, 113–123. [CrossRef] [PubMed]
- Wu, C.W.; Liu, X.; Barczyński, M.; Kim, H.Y.; Dionigi, G.; Sun, H.; Chiang, F.Y.; Kamani, D.; Randolph, G.W. Optimal stimulation during monitored thyroid surgery: EMG response characteristics in a porcine model. *Laryngoscope* 2017, 127, 998–1005. [CrossRef] [PubMed]
- Preston, D.C.; Shapiro, B.E. *Electromyography and Neuromuscular Disorders*, 4th ed.; Elsevier: Amsterdam, The Netherlands, 2020; ISBN 9780323661805.
- Papagiannis, G.I.; Triantafyllou, A.I.; Roumpelakis, I.M.; Zampeli, F.; Garyfallia Eleni, P.; Koulouvaris, P.; Papadopoulos, E.C.; Papagelopoulos, P.J.; Babis, G.C. Methodology of surface electromyography in gait analysis: Review of the literature. *J. Med. Eng. Technol.* 2019, 43, 59–65. [CrossRef]
- Hellig, T.; Johnen, L.; Mertens, A.; Nitsch, V.; Brandl, C. Investigation of observational methods assessing workload of static working postures based on surface electromyography. *Work* 2019, 62, 185–195. [CrossRef] [PubMed]
- 8. Kinali, G.; Kara, S.; Yıldırım, M.S. Electromyographic analysis of an ergonomic risk factor: Overhead work. *J. Phys. Ther. Sci.* **2016**, *28*, 1924–1927. [CrossRef]
- Motamedzade, M.; Afshari, D.; Soltanian, A.R. The Impact of Ergonomically Designed Workstations on Shoulder EMG Activity during Carpet Weaving. *Health Promot. Perspect.* 2014, 4, 144–150. [CrossRef]

- Roggio, F.; Vitale, E.; Filetti, V.; Rapisarda, V.; Musumeci, G.; Romano, E. Ergonomic Evaluation of Young Agricultural Operators Using Handle Equipment Through Electromyography and Vibrations Analysis Between the Fingers. *Saf. Health Work.* 2022, 13, 440–447. [CrossRef]
- Ranavolo, A.; Serrao, M.; Draicchio, F. Critical Issues and Imminent Challenges in the Use of sEMG in Return-To-Work Rehabilitation of Patients Affected by Neurological Disorders in the Epoch of Human–Robot Collaborative Technologies. *Front. Neurol.* 2020, 22, 572069. [CrossRef]
- Fernández-Lázaro, D.; Mielgo-Ayuso, J.; Adams, D.P.; González-Bernal, J.J.; Araque, A.F.; Cano García, A.; Fernández-Lázaro, C.I. Electromyography: A Simple and Accessible Tool to Assess Physical Performance and Health during Hypoxia Training. A Systematic Review. *Sustainability* 2020, *12*, 9137. [CrossRef]
- 13. Fukuhara, S.; Kawashima, T.; Oka, H. Indices reflecting muscle contraction performance during exercise based on a combined electromyography and mechanomyography approach. *Sci. Rep.* **2021**, *11*, 21208. [CrossRef] [PubMed]
- Ficek, K.; Gołas, A.; Pietraszewski, P.; Strózik, M.; Krzysztofik, M. The Effects of a Combined Pre- and Post-Operative Anterior Cruciate Ligament Reconstruction Rehabilitation Program on Lower Extremity Muscle Imbalance. *Appl. Sci.* 2022, *12*, 7411. [CrossRef]
- 15. Lee, K. EMG-Triggered Pedaling Training on Muscle Activation, Gait, and Motor Function for Stroke Patients. *Brain Sci.* 2022, 12, 76. [CrossRef] [PubMed]
- Osborne, J.A.; Botkin, R.; Colon-Semenza, C.; DeAngelis, T.R.; Gallardo, O.G.; Kosakowski, H.; Martello, J.; Pradhan, S.; Rafferty, M.; Readinger, J.L.; et al. Physical therapist management of parkinson disease: A clinical practice guideline from the American Physical Therapy Association. *Phys. Ther.* 2021, *102*, pzab302. [CrossRef] [PubMed]
- 17. Pilkar, R.; Momeni, K.; Ramanujam, A.; Ravi, M.; Garbarini, E.; Gail, F. Forrest; Use of Surface EMG in Clinical Rehabilitation of Individuals With SCI: Barriers and Future Considerations. *Front. Neurol.* **2020**, *11*, 578559. [CrossRef] [PubMed]
- 18. Musselman, K.E.; Shah, M.; Zariffa, J. Rehabilitation technologies and interventions for individuals with spinal cord injury: Translational potential of current trends. *J. Neuroeng. Rehabil.* **2018**, *15*, 40. [CrossRef] [PubMed]
- 19. Campanini, I.; Disselhorst-Klug, C.; Rymer, W.Z.; Merletti, R. Surface EMG in Clinical Assessment and Neurorehabilitation: Barriers Limiting Its Use. *Front. Neurol.* **2020**, *11*, 934. [CrossRef]
- 20. Cavalcanti Garciaa, M.A.; Vieira, T.M.M. Surface electromyography: Why, when and how to use it. *Rev. Andal. Med. Deporte* **2011**, *4*, 17–28.
- Ancillao, A. Analysis and Measurement of Human Motion: Modern Protocols and Clinical Considerations. J. Robot. Mech. Eng. Res. 2016, 1, 30–37. [CrossRef]
- 22. Rimington, R.P.; Fleming, J.W.; Capel, A.J.; Wheeler, P.C.; Lewis, M.P. Bioengineered model of the human motor unit with physiologically functional neuromuscular junctions. *Sci. Rep.* **2021**, *11*, 11695. [CrossRef]
- Song, S.; Lee, K.; Jung, S.; Park, S.; Cho, H.; Lee, G. Effect of Horizontal Whole-Body Vibration Training on Trunk and Lower-Extremity Muscle Tone and Activation, Balance, and Gait in a Child with Cerebral Palsy. *Am. J. Case Rep.* 2018, 19, 1292–1300. [CrossRef] [PubMed]
- Xie, T.; Leng, Y.; Zhi, Y.; Jiang, C.; Tian, N.; Luo, Z.; Yu, H.; Song, R. Increased Muscle Activity Accompanying With Decreased Complexity as Spasticity Appears: High-Density EMG-Based Case Studies on Stroke Patients. *Front. Bioeng. Biotechnol.* 2020, 16, 589321. [CrossRef] [PubMed]
- del Olmo, M.; Domingo, R. EMG Characterization and Processing in Production Engineering. *Materials* 2020, 13, 5815. [CrossRef] [PubMed]
- 26. Hakonen, M.; Piitulainen, H.; Visala, A. Current state of digital signal processing in myoelectric interfaces and related applications. *Biomed. Signal Process. Control* **2015**, *18*, 334–359. [CrossRef]
- Ye, Y.; Liu, C.; Zemiti, N.; Yang, C. Optimal Feature Selection for EMG-Based Finger Force Estimation Using LightGBM Model. In Proceeding of the 2019 28th IEEE International Conference on Robot and Human Interactive Communication (RO-MAN), New Delhi, India, 14–18 October 2019; pp. 1–7. [CrossRef]
- Alazrai, R.; Khalifeh, A.; Alnuman, N.; Alabed, D.; Mowafi, Y. An Ensemble-Based Regression Approach for Continuous Estimation of Wrist and Fingers Movements from Surface Electromyography. In Proceeding of the 38th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), Orlando, FL, USA, 16–20 August 2016. [CrossRef]
- 29. Nayak, J.; Naik, B.; Byomakesha Dash, P.; Souri, A.; Shanmuganathan, V. Hyper-parameter tuned light gradient boosting machine using memetic firefly algorithm for hand gesture recognition. *Appl. Soft Comput.* **2021**, 107, 107478. [CrossRef]
- Yang, K.; Xu, M.; Yang, X.; Yang, R.; Chen, Y. A Novel EMG-Based Hand Gesture Recognition Framework Based on Multivariate Variational Mode Decomposition. *Sensors* 2021, 21, 7002. [CrossRef] [PubMed]
- Huang, J.; Lin, S.; Wang, N.; Dai, G. TSE-CNN: A Two-Stage End-to-End CNN for Human Activity Recognition. *IEEE J. Biomed. Health Inform.* 2019, 24, 292–299. [CrossRef]
- 32. Gaetani, F.; Primiceri, P.; Zappatore, G.A.; Visconti, P. Design of an Arduino-based platform interfaced by Bluetooth Low Energy with MYO armband for controlling an under-actuated transradial prosthesis. In Proceeding of the IEEE 2018 International Conference on IC Design & Technology (ICICDT), Otranto, Italy, 4–6 June 2018; pp. 185–188. [CrossRef]
- 33. Gatewood, C.T.; Tran, A.A.; Dragoo, J.L. The efficacy of post-operative devices following knee arthroscopic surgery: A systematic review. *Knee Surg. Sport. Traumatol. Arthrosc.* 2017, 25, 501–516. [CrossRef]

- Peretti, A.; Amenta, F.; Tayebati, S.K.; Nittari, G.; Mahdi, S.S. Telerehabilitation: Review of the State-of-the-Art and Areas of Application. JMIR Rehabil. Assist. Technol. 2017, 4, e7. [CrossRef]
- Mani, S.; Sharma, S.; Omar, B.; Paungmali, A.; Joseph, L. Validity and reliability of Internet-based physiotherapy assessment for musculoskeletal disorders: A systematic review. J. Telemed. Telecare 2017, 23, 379–391. [CrossRef]
- 36. Busch, C.; Baumbach, C.; Willemsen, D.; Nee, O.; Gorath, T.; Hein, A.; Scheffold, T. Supervised training with wireless monitoring of ECG, blood pressure and oxygen-saturation in cardiac patients. *J. Telemed. Telecare* **2009**, *15*, 112–114. [CrossRef] [PubMed]
- Cottrell, M.A.; Hill, A.J.; O'Leary, S.P.; Raymer, M.E.; Russell, T.G. Service provider perceptions of telerehabilitation as an additional service delivery option within an Australian neurosurgical and orthopaedic physiotherapy screening clinic: A Qualitative Study. *Musculoskelet Sci. Pract.* 2017, 32, 7–16. [CrossRef] [PubMed]
- 38. Cottrell, M.A.; Hill, A.J.; O'Leary, S.P.; Raymer, M.E.; Russell, T.G. Clinicians' perspectives of a novel home-based multidisciplinary telehealth service for patients with chronic spinal pain. *Int. J. Telerehabilit.* **2018**, *10*, 81–88. [CrossRef] [PubMed]
- Howard, I.M.; Kaufman, M.S. Telehealth applications for outpatients with neuromuscular or musculoskeletal disorders. *Muscle Nerve* 2018, 58, 475–485. [CrossRef] [PubMed]
- 40. Eriksson, L.; Lindström, B.; Ekenberg, L. Patients' experiences of telerehabilitation at home after shoulder joint replacement. *J. Telemed. Telecare* **2011**, *17*, 25–30. [CrossRef] [PubMed]
- Jansen-Kosterink, S.; Huisin't Veld, R.; Hermens, H.; Vollenbroek-Hutten, M. A telemedicine service as partial replacement of face-to-face physical rehabilitation: The relevance of use. *Telemed. J. E-Health* 2015, 21, 808–813. [CrossRef] [PubMed]
- 42. Russell, T.G.; Buttrum, P.; Wootton, R.; Jull, G.A.; Jull, G.A. Internet-based outpatient telerehabilitation for patients following total knee arthroplasty: A randomized controlled trial. *J. Bone Jt. Surg. Am.* **2011**, *93*, 113–120. [CrossRef]
- Kairy, D.; Tousignant, M.; Leclerc, N.; Côté, A.M.; Levasseur, M.; Researchers, T.T. The patient's perspective of in-home telerehabilitation physiotherapy services following total knee arthroplasty. *Int. J. Environ. Res. Public Health.* 2013, 10, 3998–4011. [CrossRef]
- Ben-Pazi, H.; Browne, P.; Chan, P.; Cubo, E.; Guttman, M.; Hassan, A.; Hatcher-Martin, J.; Mari, Z.; Moukheiber, E.; Okubadejo, N.U.; et al. The Promise of Telemedicine for Movement Disorders: An Interdisciplinary Approach. *Curr. Neurol. Neurosci. Rep.* 2018, 18, 26. [CrossRef]
- Lewis, A.; Knight, E.; Bland, M.; Middleton, J.; Mitchell, E.; McCrum, K.; Conway, J.; Bevan-Smith, E. Feasibility of an online platform delivery of pulmonary rehabilitation for individuals with chronic respiratory disease. *BMJ Open Respir. Res.* 2021, *8*, e000880. [CrossRef]
- 46. Milani, G.; Dematte, G.; Ferioli, M.; Dallaga, G.; Lavezzi, S.; Basaglia, N.; Straudi, S. Telerehabilitation in Italy during the COVID-19 lockdown: A Feasibility and Acceptability Study. *Int. J. Telerehabil.* **2021**, *13*, e6334. [CrossRef] [PubMed]
- Eannucci, E.F.; Hazel, K.; Grundstein, M.J.; Nguyen, J.T.; Gallegro, J. Patient satisfaction for telehealth physical therapy services was comparable to that of in-person services during the COVID-19 pandemic. HSS J. Musculoskelet. J. Hosp. Spec. Surg. 2020, 16, 10–16. [CrossRef] [PubMed]
- Hongqian, X.; Fang, L.; Jingwen, N. Motion Research of Power-Assisted Lower Extremity Exoskeleton under Walking Process. Mech. Des. Manuf. 2019, 341, 107–110. [CrossRef]
- 49. Jiang, N.; Falla, D.; d'Avella, A.; Graimann, B.; Farina, D. Myoelectric control in neurorehabilitation. *Crit. Rev. Biomed. Eng.* 2010, 38, 381–391. [CrossRef] [PubMed]
- 50. Barroso, F.; Bueno, D.R.; Gallego, J.A.; Jaramillo, P.; Kilicarslan, A. Surface emg in neurorehabilitation and ergonomics: State of the art and future perspectives. *Emerg. Ther. Neurorehabilit.* **2014**, *4*, 267–284.
- Manca, A.; Cereatti, A.; Bar-On, L.; Botter, A.; Della Croce, U.; Knaflitz, M.; Maffiuletti, N.A.; Mazzoli, D.; Merlo, A.; Roatta, S.; et al. A survey on the use and barriers of surface electromyography in neurorehabilitation. *Front. Neurol.* 2020, *11*, 1137. [CrossRef] [PubMed]
- 52. Wang, J.; Dai, Y.; Si, X. Analysis and Recognition of Human Lower Limb Motions Based on Electromyography (EMG) Signals. *Electronics* **2021**, *10*, 2473. [CrossRef]
- Visconti, P.; Gaetani, F.; Zappatore, G.A.; Primiceri, P. Technical features and functionalities of MYO armband: An overview on related literature and advanced applications of myoelectric bracelets mainly focused on arm prostheses. *Int. J. Smart Sens. Intell. Syst.* 2018, 11, 1–25. [CrossRef]
- Gaetani, F.; Primiceri, P.; Zappatore, G.A.; Visconti, P. Hardware design and software development of a motion control and driving system for transradial prosthesis based on a wireless myo-electric armband. *IET Sci. Meas. Technol.* 2019, 13, 354–362. [CrossRef]
- Gordleeva, S.Y.; Lobov, S.A.; Grigorev, N.A.; Savosenkov, A.O.; Shamshin, M.O.; Lukoyanov, M.V.; Khoruzhko, M.A.; Kazantsev, V.B. Real-Time EEG–EMG Human–Machine Interface-Based Control System for a Lower-Limb Exoskeleton. *IEEE Access* 2020, *8*, 84070–84081. [CrossRef]
- 56. Stephenson, A.; Stephens, J. An exploration of physiotherapists' experiences of robotic therapy in upper limb rehabilitation within a stroke rehabilitation centre. *Disabil. Rehabil. Assist. Technol.* **2018**, *13*, 245–252. [CrossRef] [PubMed]
- 57. Lin, S.H.; Dionne, T.P. Interventions to improve movement and functional outcomes in adult stroke rehabilitation: Review and evidence summary. *J. Particip. Med.* **2018**, *10*, e3. [CrossRef] [PubMed]
- 58. Aceves-Fernandez, M.A. Artificial Intelligence: Applications in Medicine and Biology; Intech Open: London, UK, 2019; ISBN 978-1-78984-018-6. Available online: https://www.intechopen.com/books/7723 (accessed on 11 April 2022). [CrossRef]

- 59. Al-Ayyad, M.; Nazeeh, B.; Qasem, N.; Al-Takruri, M. Controlling a Lower-Leg Exoskeleton Using Voltage and Current Variation Signals of a DC Motor Mounted at the Knee Joint. *J. Med. Syst.* **2019**, *43*, 229. [CrossRef] [PubMed]
- Qian, Q.; Nam, C.; Rong, W.; Li, W.; Guo, Z.; Huang, Y.; Hu, X.; Zheng, Y.; Poon, W. Robotic and neuromuscular electrical stimulation (NMES) hybrid system. In *Intelligent Biomechatronics in Neurorehabilitation*; Academic Press: Cambridge, MA, USA, 2020; pp. 147–166.
- 61. Monte-Silva, K.; Piscitelli, D.; Norouzi-Gheidari, N.; Batalla, M.A.P.; Archambault, P.; Levin, M.F. Electromyogram-related neuromuscular electrical stimulation for restoring wrist and hand movement in poststroke hemiplegia: A systematic review and meta-analysis. *Neurorehabilit. Neural Repair* **2019**, *33*, 96–111. [CrossRef]
- 62. Hameed, H.K.; Hassan, W.Z.W.; Shafie, S.; Ahmad, S.A.; Jaafar, H. A review on surface electromyography-controlled hand robotic devices used for rehabilitation and assistance in activities of daily living. J. Prosthet. Orthot. 2020, 32, 3–13. [CrossRef]
- 63. Balasubramanian, S.; Garcia-Cossio, E.; Birbaumer, N.; Burdet, E.; Ramos-Murguialday, A. Is EMG a viable alternative to BCI for detecting movement intention in severe stroke? *IEEE Trans. Biomed. Eng.* **2018**, *65*, 2790–2797. [CrossRef] [PubMed]
- 64. Shenoy, S. EMG in sports rehabilitation. Br. J. Sport. Med. 2010, 44, i10. [CrossRef]
- 65. Felici, F.; Vecchio, A.D. Surface Electromyography: What limits its use in exercise and sport physiology? *Front. Neurol.* **2020**, *11*, 578504. [CrossRef]
- 66. de Sire, A.; Demeco, A.; Marotta, N.; Moggio, L.; Palumbo, A.; Iona, T.; Ammendolia, A. Anterior Cruciate Ligament Injury Prevention Exercises: Could a Neuromuscular Warm-Up Improve Muscle Pre-Activation before a Soccer Game? A Proof-of-Principle Study on Professional Football Players. *Appl. Sci.* 2021, *11*, 4958. [CrossRef]
- Paiva, A.; Catarino, A.; Carvalho, H.; Postolache, O.; Postolache, G.; Ferreira, F. Design of a Long Sleeve T-Shirt with ECG and EMG for Athletes and Rehabilitation Patients. In *Innovation, Engineering and Entrepreneurship. HELIX 2018. Lecture Notes in Electrical Engineering*; Machado, J., Soares, F., Veiga, G., Eds.; Springer: Cham, Switzerland, 2018; Volume 505, ISBN 978-3-319-91333-9. [CrossRef]
- Smeets, A.; Verschueren, S.; Staes, F.; Vandenneucker, H.; Claes, S.; Vanrenterghem, J. Athletes with an ACL reconstruction show a different neuromuscular response to environmental challenges compared to uninjured athletes. *Gait Posture* 2021, *83*, 44–51. [CrossRef]
- 69. Zebis, M.K.; Andersen, L.L.; Brandt, M.; Myklebust, G.; Bencke, J.; Bloch Lauridsen, H.; Bandholm, T.; Thorborg, K.; Hölmich, P.; Aagaard, P. Effects of evidence-based prevention training on neuromuscular and biomechanical risk factors for ACL injury in adolescent female athletes: A randomised controlled trial. *Br. J. Sport. Med.* **2016**, *50*, *552*–557. [CrossRef] [PubMed]
- 70. Sun, J.; Liu, G.; Sun, Y.; Lin, K.; Zhou, Z.; Cai, J. Application of Surface Electromyography in Exercise Fatigue: A Review. *Front. Syst. Neurosci.* **2022**, *16*, 893275. [CrossRef] [PubMed]
- 71. Rampichini, S.; Martins Vieira, T.; Castiglioni, P.; Merati, G. Complexity Analysis of Surface Electromyography for Assessing the Myoelectric Manifestation of Muscle Fatigue: A Review. *Entropy* **2020**, *22*, 529. [CrossRef] [PubMed]
- 72. Massó, N.; Rey, F.; Romero, D.; Gual, G.; Costa, L.; Germán, A. Surface electromyography applications in the sport. *Apunt. Med. L'esport* **2010**, *45*, 121–130.
- 73. Hussain, J.; Sundaraj, K.; Low, Y.F.; Kiang, L.C.; Sundaraj, S.; Ali, M.A. A systematic review on fatigue analysis in triceps brachii using surface electromyography. *Biomed. Signal Process. Control* **2018**, *40*, 396–414. [CrossRef]
- 74. Sgroi, T.A.; Cilenti, M. Rotator cuff repair: Post-operative rehabilitation concepts. *Curr. Rev. Musculoskelet. Med.* **2018**, *11*, 86–91. [CrossRef] [PubMed]
- 75. Edwards, P.K.; Ebert, J.R.; Littlewood, C.; Ackland, T.; Wang, A. A systematic review of electromyography studies in normal shoulders to inform post-operative rehabilitation following rotator cuff repair. *J. Orthop. Sport. Phys. Ther.* **2017**, 47, 931–944. [CrossRef]
- Scott, B.R.; Slattery, K.M.; Sculley, D.V.; Lockhart, C.; Dascombe, B.J. Acute Physiological Responses to Moderate-Load Resistance Exercise in Hypoxia. J. Strength Cond. Res. 2017, 31, 1973–1981. [CrossRef]
- 77. Taborri, J.; Keogh, J.; Kos, A.; Santuz, A.; Umek, A.; Urbanczyk, C.; van der Kruk, E.; Rossi, S. Sport Biomechanics Applications Using Inertial, Force, and EMG Sensors: A Literature Overview. *Appl. Bionics Biomech.* **2020**, 2020, 2041549. [CrossRef]
- 78. Kos, A.; Umek, A. Wearable sensor devices for prevention and rehabilitation in healthcare: Swimming exercise with real-time therapist feedback. *IEEE Internet Things J.* 2019, *6*, 1331–1341. [CrossRef]
- 79. Jiang, Y. Combination of wearable sensors and internet of things and its application in sports rehabilitation. *Comput. Commun.* **2020**, *150*, 167–176. [CrossRef]
- 80. Lynn, S.K.; Watkins, C.; Wong, M.A.; Balfany, K.; Feeney, D.F. Validity and reliability of surface electromyography measurements from a wearable athlete performance system. *J. Sport. Sci. Med.* **2018**, *17*, 205.
- Rajeswari, J.; Jagannath, M. Advances in biomedical signal and image processing—A systematic review. *Inform. Med. Unlocked* 2017, 8, 13–19. [CrossRef]
- 82. Gohel, V.; Mehendale, N. Review on electromyography signal acquisition and processing. *Biophys. Rev.* 2020, *12*, 1361–1367. [CrossRef]
- 83. Houssein, E.H.; Kilany, M.; Hassanien, A.E. ECG signals classification: A review. Int. J. Intell. Eng. Inform. 2017, 5, 376–396. [CrossRef]
- 84. Faust, O.; Hagiwara, Y.; Hong, T.J.; Lih, O.S.; Acharya, U.R. Deep learning for healthcare applications based on physiological signals: A review. *Comput. Methods Programs Biomed.* **2018**, *161*, 1–13. [CrossRef] [PubMed]

- Jull, G.; Moore, A.; Falla, D.; Lewis, J.; McCarthy, C.; Sterling, M. Grieve's Modern Musculoskeletal Physiotherapy, 4th ed.; Elsevier Health Sciences: Amsterdam, The Netherlands, 2015; ISBN 9780702051524.
- 86. Rahul, J.; Sora, M.; Sharma, L.D. An overview on biomedical signal analysis. Int. J. Recent Technol. Eng. 2019, 7, 206–209.
- 87. Tankisi, H.; Burke, D.; Cui, L.; de Carvalho, M.; Kuwabara, S.; Nandedkar, S.D.; Rutkove, S.; Stalberg, E.; van Putten, M.J.A.M.; Fuglsang-Frederiksen, A. Standards of instrumentation of EMG. *Clin. Neurophysiol.* **2020**, *131*, 243–258. [CrossRef]
- 88. Kimura, J. *Electrodiagnosis in Diseases of Nerve and Muscle: Principles and Practice*, 4th ed.; Oxford University Press: New York, NY, USA, 2013; ISBN 9780199969296.
- 89. Cuesta-Vargas, A. EMG Analysis of the Neuromuscular Activity during Sit-to-Stand from Different Height Chairs in Water. *Int. J. Aquat. Res. Educ.* 2019, 12, 6. [CrossRef]
- 90. Chan, A.Y.K. *Biomedical Device Technology: Principles and Design*, 2nd ed.; Charles, C., Ed.; Thomas Publisher, Limited: Springfield, IL, USA, 2016; ISBN 9780398090838.
- Tun, N.N.; Sanuki, F.; Iramina, K. Electroencephalogram Electromyogram Functional Coupling and Delay Time Change Based on Motor Task Performance. *Sensors* 2021, 21, 4380. [CrossRef]
- 92. Caldwell, R.R.; Lovely, D.F. Commercial Hardware for the Implementation of Myoelectric Control. In *Powered Upper Limb Prostheses*; Muzumdar, A., Ed.; Springer: Berlin/Heidelberg, Germany, 2004; pp. 55–71. [CrossRef]
- Huang, Z.-X.; Zhang, X.-D.; Li, Y.-N. Design of a grasp force adaptive control system with tactile and slip perception. In Proceedings of the 2012 IEEE International Conference on Automation Science and Engineering (CASE), Seoul, Republic of Korea, 20–24 August 2012; pp. 1101–1105. [CrossRef]
- 94. Merletti, R.; Farina, D. Surface Electromyography: Physiology, Engineering, and Application, 1st ed.; Wiley-IEEE Press Series on Biomedical Engineering: Hoboken, NJ, USA, 2016; ISBN 9781118987025. [CrossRef]
- 95. McManus, L.; De Vito, G.; Lowery, M.M. Analysis and biophysics of surface EMG for physiotherapists and kinesiologists: Toward a common language with rehabilitation engineers. *Front. Neurol.* **2020**, *11*, 576729. [CrossRef] [PubMed]
- 96. Gaetani, F.; De Fazio, R.; Zappatore, G.A.; Visconti, P. A Prosthetic Limb Managed by Sensors-Based Electronic System: Experimental Results on Amputees. *Bull. Electr. Eng. Inform. BEEI* **2020**, *9*, 314–324. [CrossRef]
- Campanini, I.; Merlo, A.; Disselhorst-Klug, C.; Mesin, L.; Muceli, S.; Merletti, R. Fundamental Concepts of Bipolar and High-Density Surface EMG Understanding and Teaching for Clinical, Occupational, and Sport Applications: Origin, Detection, and Main Errors. *Sensors* 2022, 22, 4150. [CrossRef] [PubMed]
- Merlo, A.; Bò, M.C.; Campanini, I. Electrode Size and Placement for Surface EMG Bipolar Detection from the Brachioradialis Muscle: A Scoping Review. Sensors 2021, 21, 7322. [CrossRef]
- 99. Péter, A.; Andersson, E.; Hegyi, A.; Finni, T.; Tarassova, O.; Cronin, N.; Grundström, H.; Arndt, A. Comparing Surface and Fine-Wire Electromyography Activity of Lower Leg Muscles at Different Walking Speeds. *Front. Physiol.* **2019**, *10*, 1283. [CrossRef]
- Farina, D.; Merletti, R.; Indino, B.; Nazzaro, M.; Pozzo, M. Surface EMG Crosstalk between Knee Extensor Muscles: Experimental and Model Results. *Muscle Nerve* 2002, 26, 681–695. [CrossRef]
- 101. Meng, Q.; Zhang, J.; Yang, X. Virtual rehabilitation training system based on surface EMG feature extraction and analysis. *J. Med. Syst.* **2019**, *43*, 48. [CrossRef]
- 102. Moore, J.; Zouridakis, G. *Biomedical Technology and Devices Handbook*; The Mechanical Engineering Handbook Series; CRC Press: Boca Raton, FL, USA, 2004.
- Song, H.; Park, Y.; Kim, H.; Ko, H. Fully integrated biopotential acquisition analog front-end IC. Sensors 2015, 15, 25139–25156.
 [CrossRef]
- 104. Maceira-Elvira, P.; Popa, T.; Schmid, A.C.; Hummel, F.C. Wearable technology in stroke rehabilitation: Towards improved diagnosis and treatment of upper-limb motor impairment. *J. Neuroeng. Rehabil.* **2019**, *16*, 142. [CrossRef]
- Porciuncula, F.; Roto, A.V.; Kumar, D.; Davis, I.; Roy, S.; Walsh, C.J.; Awad, L.N. Wearable movement sensors for rehabilitation: A focused review of technological and clinical advances. J. Inj. Funct. Rehabil. 2018, 10 (Suppl. S2), S220–S232. [CrossRef]
- 106. Rodgers, M.M.; Alon, G.; Pai, V.M.; Conroy, R.S. Wearable technologies for active living and rehabilitation: Current research challenges and future opportunities. *J. Rehabil. Assist. Technol. Eng.* **2019**, *6*, 2055668319839607. [CrossRef] [PubMed]
- 107. Li, C.; Un, K.F.; Mak, P.I.; Chen, Y.; Muñoz-Ferreras, J.M.; Yang, Z.; Gómez-García, R. Overview of recent development on wireless sensing circuits and systems for healthcare and biomedical applications. *IEEE J. Emerg. Sel. Top. Circuits Syst.* 2018, *8*, 165–177. [CrossRef]
- Baig, M.M.; Gholam Hosseini, H.; Moqeem, A.A.; Mirza, F.; Lindén, M. A systematic review of wearable patient monitoring systems—Current challenges and opportunities for clinical adoption. J. Med. Syst. 2017, 41, 115. [CrossRef] [PubMed]
- 109. Kumari, P.; Mathew, L.; Syal, P. Increasing trend of wearables and multi-modal interface for human activity monitoring: A review. *Biosens. Bioelectron.* **2017**, *90*, 298–307. [CrossRef] [PubMed]
- 110. Zhao, S.; Liu, J.; Gong, Z.; Lei, Y.; OuYang, X.; Chan, C.C.; Ruan, S. Wearable Physiological Monitoring System Based on Electrocardiography and Electromyography for Upper Limb Rehabilitation Training. *Sensors* **2020**, *20*, 4861. [CrossRef]
- 111. Erickson, J.C.; Stepanyan, E.; Hassid, E. Comparison of Dry and Wet Electrodes for Detecting Gastrointestinal Activity Patterns from Body Surface Electrical Recordings. *Ann. Biomed. Eng.* **2023**, 1–12. [CrossRef]
- Liu, S.-H.; Wang, J.-J.; Tan, T.-H. A Portable and Wireless Multi-Channel Acquisition System for Physiological Signal Measurements. Sensors 2019, 19, 5314. [CrossRef]

- Nguyen, M.; Gia, T.N.; Westerlund, T. EMG-Based IoT System Using Hand Gestures for Remote Control Applications. In Proceedings of the 2021 IEEE 7th World Forum on Internet of Things (WF-IoT), New Orleans, LA, USA, 14 June–31 July 2021; pp. 911–912.
- Park, S.Y.; Cho, J.; Na, K.; Yoon, E. Modular 128-Channel Δ-ΔΣ Analog Front-End Architecture Using Spectrum Equalization Scheme for 1024-Channel 3-D Neural Recording Microsystems. *IEEE J.-Solid-State Circuits* 2017, 53, 501–514. [CrossRef]
- Kast, C.; Krenn, M.; Aramphianlert, W.; Hofer, C.; Aszmann, O.C.; Mayr, W. Modular Multi-channel Real-time Bio-signal Acquisition System. In Proceedings of the International Conference on Advancements of Medicine and Health Care through Technology, Cluj-Napoca, Romania, 12–15 October 2016; pp. 95–98.
- 116. Tran, L.; Cha, H.K. An ultra-low-power neural signal acquisition analog front-end IC. Microelectron. J. 2021, 107, 104950. [CrossRef]
- Piccinini, D.J.; Andino, N.B.; Ponce, S.D.; Roberti, M.A. Wearable system for acquisition and monitoring of biological signals. J. Phys. Conf. Ser. 2016, 705, 012009. [CrossRef]
- Sarker, V.K.; Jiang, M.; Gia, T.N.; Anzanpour, A.; Rahmani, A.M.; Liljeberg, P. Portable multipurpose bio-signal acquisition and wireless streaming device for wearables. In Proceedings of the 2017 IEEE Sensors Applications Symposium (SAS), Glassboro, NJ, USA, 13–15 March 2017; pp. 1–6.
- 119. Mazzetta, I.; Gentile, P.; Pessione, M.; Suppa, A.; Zampogna, A.; Bianchini, E.; Irrera, F. Stand-alone wearable system for ubiquitous real-time monitoring of muscle activation potentials. *Sensors* **2018**, *18*, 1748. [CrossRef] [PubMed]
- 120. Zhang, R.; Amft, O. Bite Glasses: Measuring Chewing Using Emg and Bone Vibration in Smart Eyeglasses. In Proceedings of the 2016 ACM International Symposium on Wearable Computers, Heidelberg, Germany, 12–16 September 2016; pp. 50–52.
- 121. Kim, I.; Bhagat, Y.A.; Homer, J.; Lobo, R. Multimodal analog front end for wearable bio-sensors. *IEEE Sens. J.* **2016**, *16*, 8784–8791. [CrossRef]
- 122. Nakamura, H.; Sakajiri, Y.; Ishigami, H.; Ueno, A. A Novel Analog Front End with Voltage-Dependent Input Impedance and Bandpass Amplification for Capacitive Biopotential Measurements. *Sensors* 2020, 20, 2476. [CrossRef] [PubMed]
- 123. Biagetti, G.; Crippa, P.; Falaschetti, L.; Orcioni, S.; Turchetti, C. Human activity monitoring system based on wearable sEMG and accelerometer wireless sensor nodes. *Biomed. Eng. Online* **2018**, *17*, 132. [CrossRef]
- 124. Biagetti, G.; Crippa, P.; Falaschetti, L.; Turchetti, C. A multi-channel electromyography, electrocardiography and inertial wireless sensor module using Bluetooth low-energy. *Electronics* **2020**, *9*, 934. [CrossRef]
- Li, X.; Sun, Y. NCMB-button: A wearable non-contact system for long-term multiple biopotential monitoring. In Proceedings of the 2017 IEEE/ACM International Conference on Connected Health: Applications, Systems and Engineering Technologies (CHASE), Philadelphia, PA, USA, 17–19 July 2017; pp. 348–355.
- 126. Velumani, D.; Nancy Subha, G.; Harishma, S.; Kruthika Reshmi, C.; Meghana, P.; Gokul, M. IoT Based Tele-Sleep Bruxism Diagnosis Tool. In Proceedings of the 2022 International Conference on Augmented Intelligence and Sustainable Systems (ICAISS), Trichy, India, 24–26 November 2022; pp. 1156–1160.
- 127. Senapati, B.; Kumar, M.G.L.; Ray, K.B. High resolution reconfigurable bio-potential processor for portable biomedical application. In *Devices for Integrated Circuit (DevIC)*; IEEE: New York, NY, USA, 2017; pp. 517–521.
- 128. Lee, S.C.; Lin, Y.S.; Chen, Y.J.; Chiueh, H. A wireless multi-channel physiological signal acquisition system-on-chip for wearable devices. In Proceedings of the 2016 IEEE Sensors Conference, Orlando, FL, USA, 30 October–3 November 2016. [CrossRef]
- 129. Augustyniak, P. Remotely programmable architecture of a multi-purpose physiological recorder. *Microprocess. Microsyst.* **2016**, *46*, 55–66. [CrossRef]
- 130. Bhamra, H.; Lynch, J.; Ward, M.; Irazoqui, P. A noise-power-area optimized biosensing front end for wireless body sensor nodes and medical implantable devices. *IEEE Trans. Very Large Scale Integr. (VLSI) Syst.* 2017, 25, 2917–2928. [CrossRef]
- 131. Dai, Y.; Jin Wu, J.; Niu, J.; Gu, F.; Shen, S. MSEva: A Musculoskeletal Rehabilitation Evaluation System Based on EMG Signals. *ACM Trans. Sens. Netw.* **2022**, *19*, 6. [CrossRef]
- 132. De Fazio, R.; De Vittorio, M.; Visconti, P. Innovative IoT Solutions and Wearable Sensing Systems for Monitoring Human Biophysical Parameters: A Review. *Electronics* **2021**, *10*, 1660. [CrossRef]
- 133. De Fazio, R.; Stabile, M.; De Vittorio, M.; Velázquez, R.; Visconti, P. An Overview of Wearable Piezoresistive and Inertial Sensors for Respiration Rate Monitoring. *Electronics* **2021**, *10*, 2178. [CrossRef]
- 134. Data Acquisition Systems-Biometrics Ltd. Available online: https://www.biometricsltd.com/systems-wireless.htm (accessed on 1 September 2022).
- 135. Surface EMG Sensors-Biometrics Ltd. Available online: https://www.biometricsltd.com/surface-emg-sensor.htm (accessed on 15 August 2022).
- Wearable Sensor Products-Individual Sensors, Shimmer Co. Available online: http://www.shimmersensing.com/products/ individual-sensors/ (accessed on 4 April 2022).
- 137. Shimmer3 EMG Unit, Shimmer Co. Available online: http://www.shimmersensing.com/products/shimmer3-emg-sensor# related-tab (accessed on 4 April 2022).
- 138. Shimmer3 ECG Unit, Shimmer Co. Available online: http://www.shimmersensing.com/products/shimmer3-ecg-sensor (accessed on 10 July 2022).
- González-Mendoza, A.; Pérez-SanPablo, A.I.; López-Gutiérrez, R.; Quiñones-Urióstegui, I. Validation of an EMG Sensor for Internet of Things and Robotics. In Proceedings of the 2018 15th International Conference on Electrical Engineering, Computing Science and Automatic Control (CCE), Mexico City, Mexico, 5–7 September 2018; pp. 1–5.

- Sakamoto, A.; Khan, M.T.I.; Kurita, T. EMG Signals in Co-Activations of Lower Limb Muscles for Knee Joint Analysis. In Proceedings of the 2015 International Conference on Informatics, Electronics & Vision (ICIEV), Fukuoka, Japan, 15–18 June 2015; pp. 1–5.
- 141. BioSemi Products Website, Active High-Density EMG Electrode Array—BioSemi Co. Available online: https://www.biosemi. com/HD_electrode.htm (accessed on 18 July 2022).
- 142. Angeli, T.R.; O'Grady, G.; Erickson, J.C.; Du, P.; Paskaranandavadivel, N.; Bissett, I.P.; Cheng, L.K.; Pullan, A.J. Mapping Small Intestine Bioelectrical Activity Using High-Resolution Printed-Circuit-Board Electrodes. In Proceedings of the 2011 Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Boston, MA, USA, 30 August–3 September 2011; pp. 4951–4954.
- 143. BTSbioengineering FreeEMG | Wireless Surface EMG | BTS Bioengineering. Available online: https://www.btsbioengineering. com/products/freeemg/ (accessed on 12 December 2022).
- 144. Tesio, L.; Malloggi, C.; Malfitano, C.; Coccetta, C.A.; Catino, L.; Rota, V. Limping on Split-Belt Treadmills Implies Opposite Kinematic and Dynamic Lower Limb Asymmetries. *Int. J. Rehabil. Res.* **2018**, *41*, 304–315. [CrossRef] [PubMed]
- 145. Plux Biosignals. Available online: https://www.pluxbiosignals.com/products/ (accessed on 1 July 2022).
- 146. Electromyography (EMG) Sensor—BITalino. Available online: https://www.pluxbiosignals.com/collections/myology/ products/electromyography-emg-sensor (accessed on 4 August 2022).
- 147. Plux Biosignalsplux. Available online: https://www.pluxbiosignals.com/collections/biosignalsplux (accessed on 4 August 2022).
- 148. Santos, P.; Quaresma, C.; Garcia, I.; Quintão, C. Neuromotor Evaluation of the Upper Limb during Activities of Daily Living: A Pilot Study. In Proceedings of the Technological Innovation for Digitalization and Virtualization, Cham, Switzerland, 29 June–1 July 2022; pp. 112–121.
- 149. Delsys Trigno Research+. Available online: https://delsys.com/trigno/research/ (accessed on 4 April 2022).
- Hu, Z.; Kim, Y.; Zhang, Y.; Zhang, Y.; Li, J.; Tang, X.; Sohn, J.; Kim, S. Correlation of Lower Limb Muscle Activity with Knee Joint Kinematics and Kinetics during Badminton Landing Tasks. *Int. J. Environ. Res. Public Health* 2022, 19, 16587. [CrossRef] [PubMed]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.